

# FAQ: How do I extract the output gap?

Fabio Canova

January 2020

#### WORKING PAPERS ARE OBTAINABLE FROM

www.riksbank.se/en/research

Sveriges Riksbank • SE-103 37 Stockholm Fax international: +46 8 21 05 31 Telephone international: +46 8 787 00 00

The Working Paper series presents reports on matters in the sphere of activities of the Riksbank that are considered to be of interest to a wider public. The papers are to be regarded as reports on ongoing studies and the authors will be pleased to receive comments.

The opinions expressed in this article are the sole responsibility of the author(s) and should not be interpreted as reflecting the views of Sveriges Riksbank.

### FAQ: How do I extract the output gap?

Fabio Canova \*

Norwegian Business School, CAMP and CEPR

Sveriges Riksbank Working Paper Series No. 386

January 2020

#### Abstract

I study potentials and gaps, permanent and transitory fluctuations in macroeconomic variables using the Smets and Wouter (2007) model. Model-based gaps display low frequency variations; possess more than business cycle fluctuations; have similar frequency representation as potentials, and are correlated with them. Permanent and transitory fluctuations display similar features, but are uncorrelated. I use a number of filters to extract trends and cycles using simulated data. Gaps are best approximated with a polynomial filter; transitory fluctuations with a differencing approach, but distortions are large. Explanations for the results are given. I propose a filter which reduces the biases of existing procedures.

Key words: Gaps and potentials, permanent and transitory components, filtering, cyclical fluctuations, gain functions.

JEL Classification: C31, E27, E32.

<sup>\*</sup>I thank Paolo Gelain, Jesper Linde' and the participants of seminars at the Riksbank, Bank of Finland, ECB, University of Cardiff, First MacroDolomiti Conference, Castelrotto, and the WGEM workshop, Paris for comments and suggestions. The opinions expressed in this article are the sole responsibility of the author and should not be interpreted as reflecting views of Sveriges Riksbank.

#### **1** INTRODUCTION

Since the 2008 financial crisis, academic economists and policymakers have been keenly interested in the level of certain latent variables (such the output gap, the natural rate of interest, the NAIRU, etc.) and in the changes that the crisis has brought about in nature of the cyclical fluctuations in inflation and unemployment. Unfortunately, while the profession agrees on the centrality of these issues, in practice, different users draw conclusions looking at different quantities. For example, the term output gap is interchangeably employed to refer to the difference between the actual output and its potential, defined as the level prevailing absent nominal frictions; between actual output and its permanent component; or between actual output and its (statistical long-run) trend. However, potential output may not be trending (in a statistical sense), and may feature both permanent and transitory swings. Similarly, the natural rate of interest is, at times, defined with reference to the frictions present in a model and, at other times, as the long run component of real interest rates. There are also a number of competing definition of cyclical fluctuations, see e.g. Pagan [2019], and different researchers use different approaches to extract them. Clearly, without a consensus on what the objects of interest are, the measurement of latent quantities becomes elusive. However, even if a consensus could be reached, the available tools are statistical in nature and do not generally employ, even in a reduced form sense, the information provided by the structural models economists use to discuss the features of latent variables. Hence, economic analyses and policy prescriptions may become whimsical.

Perhaps unsurprisingly, the lapse between theory and measurement creates confusion in the profession. In recent years macroeconomists have argued about what the data tells us about, e.g., potential output (see Coibon, Gorodnichenko, and Ulate [2018]), the long run properties of the natural rate of interest (see Laubach and William [2015]), or the dynamics of NAIRU (see Crump, Giannoni, and Sahini [2019]); what is the best tool to extract cyclical fluctuations (see Hamilton [2018]); whether permanent or transitory disturbances are responsible for macroeconomic fluctuations (see Schmitt-Grohe and Uribe [2019]); or which theory is consistent with business cycle facts (see Angeletos, Collard, and Dellas [2019]). Canova [1998] and Canova [1999] demonstrated that statistical methods used to separate one observable variable into two latent components (trend and cycle, for short) produce time series with different properties. Thus, without a firm stand on what the features of the objects of interest are and a reliable mapping between theoretical and statistical quantities, it is impossible to fruitfully select an extraction method for applied exercises. Canova [2014] reiterated the argument by showing that fitting stationary structural models to the output of standard statistical tools results in heterogeneous estimates of the structural parameters and in different dynamics in response to structural disturbances, both of which make inference difficult. More recently, Beaudry, Galizia, and Portier [2018] showed that there are interesting fluctuations in hours that standard filtering approaches disregard, see also Lubik, Matthes, and Verona [2019]. They also argue that these fluctuations could help researchers to understand better the type of models which are consistent with the data; see Kulish and Pagan [2019] for a critical view.

This paper attempts to shed light on the relationship between theoretical notions of gaps (transitory fluctuations) and the cycle one recovers with statistical approaches using a laboratory economy. I conduct a Monte Carlo exercise employing a version of the Smets and Wouters [2007] model as the data generating process (DGP). This model seems a natural starting point for the exercise for three reasons: it has a good fit to the data of many countries; it has been used to analyze policy trade-offs and optimal monetary policy decisions, see e.g. Justiniano, Primiceri, and Tambalotti [2013]; many policy institutions use versions of this model for their out-of-sample forecasting exercises.

In such a model, latent variables are well defined objects. Potentials are the equilibrium outcomes obtained eliminating nominal frictions, markup and monetary disturbances and the gaps are the deviations between the level variables and the potential outcomes. Similarly, the permanent component is what the model produces when certain disturbances have permanent features, while the transitory component is the difference between the level variables and the permanent component. With this taxonomy in mind, one can evaluate which statistical approach produces estimates which are close to those of model-based gaps (transitory components) and examine the reasons for why distortions occur. While the quantitative results I present are clearly model-dependent, the features responsible for the distortions are independent of the details of the lab economy. In particular the calibration of the experiment; the number or the properties of the disturbances; the presence of financial frictions, or the underlying principles used to construct the equations of the model do not affect the conclusions. I show instead that it is the interaction between the spectral properties of the latent variables and the characteristic of filters which accounts for the outcomes I obtain. In particular, the fact that gaps and potentials (transitory and permanent components) have similar spectral features, and similar distribution of variance by frequency renders standard approaches incapable of consistently estimating the latent variable of interest.

I employ numerous filters in the exercise covering well the procedures most commonly used in practice. Eight approaches are univariate (polynomial, Hodrick and Prescott, first order and long order differencing, unobservable component, band pass, wavelet, Hamilton local projection); two are multivariate (Beveridge and Nelson, Blanchard and Quah). Because many filters have free parameters, I also examine whether the ranking change when they are are set at different values.

The conclusions I obtain are somewhat surprising. When the object of investigation are the gaps, polynomial filtering minimizes the distortions relative to model-based quantities on average, across series and statistics. Because gaps and potentials display similar low frequencies components and similar distribution of spectral variability, polynomial filtering dominates because it leaves almost undistorted the distribution of spectral variability of the data while all other filters heavily change it. This conclusion is independent of the sample size and of the settings of the free parameters. When the transitory components are the object of investigation, the conclusions are less firm. Polynomial filtering is still performing well, on average across variables and statistics, in large but not in small samples. Still, simple approaches, such as first or long order differencing, are doing better than more elaborate local projection or unobservable component procedures, both in large and small samples. Also in this case, the assignment of low and business cycle frequency variations is crucial to explain the results. Procedures which simultaneously account for the fact that, at low frequency the transitory component matters and at business cycle frequencies the permanent component is important, and that the transitory component has more power at low than at business cycle frequencies come closest in capturing the dynamic features of model-based transitory components.

The presence of important low frequency fluctuations in the theory-based gaps (transitory components) and of business cycle fluctuations in the theory-based potentials (permanent components) indicates that the standard practice of defining cyclical those fluctuations with 2-8 years periodicity is bound to produce inferential distortions. Filters focusing on these fluctuations will overestimate gaps variance at business cycle frequencies and underestimating gaps variance at low frequencies, altering the sequence and the number of turning points, and the properties of expansions and recessions. It also suggests that, when disentangling gaps from potentials (transitory from permanent fluctuations), econometrically fancier procedures will not perform better than crude ones, because they are not designed to capture the theoretical features of my DGP.

Given the distortions that most procedures display, one may be tempted to go structural, estimate the model assumed to have generated the data, and with the parameter estimates construct modelbased estimates of the latent components, much in the spirit of Justiniano et al. [2013] or Furlanetto, Gelain, and Taheri-Sanjani [2020]. If model misspecification is a concern, the composite posterior approach of Canova and Matthes [2018] could be used to render latent variables measurement more robust. While structural models are popular in academics, policy institutions still use a variety of statistical methods to extract gaps (transitory components), benchmarking the outcomes with an estimated model, see e.g. Croitorov, Hristov, McMorrow, Pfeiffer, Roeger, and Vandermuellen [2019].

For this reason, I design a filter displaying some optimality properties, given the features of the gaps I emphasize. I show that the filter can be rigged to produce gaps with important low frequency variations and potentials with interesting business cycle frequencies variations; can generate time paths that are close to those of the DGP; and can improve on the best procedures in terms of the Monte Carlo statistics I construct. While the gains are more moderate when transitory components are of interest, the filter is competitive with the best existing approaches even in this case.

The rest of the paper is organized as follows. The next section provides a refresher of the terminology used in the paper and definitions needed to understand what will come next; section 3 discusses the design of the experiment; section 4 lists the filtering procedures; section 5 presents the statistics; and section 6 summarizes of outcomes. Section 7 interprets the results and designs a filter which can capture relevant features of the data. Section 8 concludes.

#### 2 A Refresher of the terminology and some definitions

A zero mean, stationary time series  $X_t$  can be equivalently characterized with the autocovariance function  $\gamma(\tau) = E(X_t X'_{t-\tau})$  or with the spectral density  $S(\omega) = \frac{1}{2\pi} \sum_{\tau=-\infty}^{\infty} e^{-i\omega\tau} \gamma(\tau), \omega = [-\pi, \pi],$ see e.g. Canova [2007]. While both functions are symmetric (around  $\tau = 0$  or  $\omega = 0$ ), the elements of the former are correlated while those of the latter are not. When the spectral density is evaluated at Fourier frequencies  $\omega_j = \frac{2\pi j}{T}$ , it is possible to associate spectral frequencies with periods of oscillations  $p = \frac{2\pi}{\omega_j}$ , and thus split the variance of the process (the area under the spectral density,  $\sum_{\omega_j} S(\omega_j)$ ) into orthogonal regions comprising cycles with different periodicity.

In time series it is typical to associate low frequencies variations, i.e. fluctuations with long period of oscillations, with trends; medium frequency variations, i.e. fluctuations with medium period of oscillations, with business cycles; and high frequency variations, i.e. fluctuations with short period of oscillations, with irregular cycles. To analyze "trends" or the "business cycles' in isolation, one typically employs filters. Ideally, a filter eliminates fluctuations at "uninteresting" frequencies while leaving unchanged the fluctuations at "interesting" frequencies. Low pass filters, i.e. filters with the frequency representation  $F(\omega_j) = 1$ , if  $\omega_j < \bar{\omega}_1$  and zero otherwise, and band pass filters, i.e. filters with the frequency representation  $F(\omega_j) = 1$ , if  $\bar{\omega}_1 < \omega_j < \bar{\omega}_2$  and zero otherwise, have these features. However, with a finite amount of data, both types of filters generate distortions due leakages over other frequencies and compressions at the required frequencies - think of a badly tuned receiver: the signal from your favorite station will be weaker than otherwise; and signals from other stations will make the reception noisy.

Apart from low pass and band pass filters, there are a variety of statistical approaches one can use to extract trends or business cycles fluctuations. Generally speaking, the procedures fall into two classes: moving average and regression methods. Because all approaches will imperfectly extract the variance of the spectrum at the required frequencies, even in large samples, they should be considered approximations to ideal filters.

The class of moving average filters is of interest among applied investigators because, if the weights sum to one, they will detrend a series, in the sense that they produce a stationary output from a non-stationary input. Because not all filters can be represented via moving averages and because the weights do not necessarily sum one, detrending and filtering are distinct operations. Furthermore, because different approaches produce approximation errors with different properties, one should expect filtered data to display different time series profiles and different moments (see e.g. Canova [1998], Canova [1999]).

When X displays a unit root,  $S(\omega_j = 0)$  goes to infinity, and the variance of X also goes to infinity. However, away from the zero frequency, one can still examine the spectral properties of non-stationary data. In particular, one can still analyze long cycles (say, those with periodicity 32 to, say, 64 quarters), provided that  $X_t$  is long enough to contain reliable information about them. While there are ways to analyze the very low frequency components of non-stationary series (for example, using the local spectrum or the growth rate of the data), when the DGP is non-stationary I will present results simply omitting the zero and neighboring frequencies (from 64 quarters to infinity).

The squared gain function provides a useful tool to understand the time series behavior of the filtered series and to interpret the distortions different filters generate. The function tells us, frequency by frequency, the proportion of the variance of the filtered series to the variance of the original series. For example, a unitary squared gain at  $\omega_j$  means that a method has left untouched the variability of the original series, while a zero squared gain means that it has wiped out all the variability at  $\omega_j$ . Squared gain values between zero and one, on the other hand, indicate the extent of the attenuation of the variability at frequency  $\omega_j$  and values in excess of one provide evidence that the filter has amplified the variance of the original series at  $\omega_j$ .

#### 3 The design of the experiment

As DGP for the experiments, I use the standard closed economy new-Keynesian model popularized by Smets and Wouters [2007] with real frictions (habit in consumption, investment adjustment costs), nominal frictions (price and wage stickiness and indexation), and a Taylor rule for interest rate determination. In a new-Keynesian setup, a set of equations characterizing the optimality conditions for the potential economy - the economy without nominal frictions, markups and monetary disturbances - is added to the optimality conditions of the original problem (see appendix) and the solution for level and potential variables is jointly found. Gaps are obtained as the difference between the level and the potential for each endogenous variable. One can assume that all the exogenous disturbances are transitory or that some of them are permanent and others are transitory. In the latter case, apart from a potential-gap decomposition, one can separate the permanent and the transitory component of the endogenous variables, the former being the portion of the data driven by permanent disturbances, and the latter the portion of the data driven by transitory disturbances. As discussed below, when some of the disturbances are permanent, potentials and gaps will both display permanent and transitory features. Thus, gaps may not be interesting economic objects (gaps may "never close"), but the transitory components of the data are.

The model economy features seven structural disturbances (to TFP, to investment, to government expenditure, to the Taylor rule, to the price and wage markups, and to the risk premium). In the baseline specification, all disturbances are stationary. In this case, potentials and gaps are stationary, there is no long run trend (permanent path) to speak of. Thus, one may wonder what is the purpose of trend-cycle decompositions in this situation. When disturbances are stationary but persistent, important low frequency variations appears in the data. Thus, to focus attention on fluctuations with interesting periodicity, a researcher may want to purge the data of low frequency variations to highlight the object of interest. Hence, even in this case, the exercises I run are relevant.

Because in the baseline specification disturbances are persistent, gaps and potentials will both feature low and business cycle fluctuations, and their spectral power at low and business cycle frequencies will generally be similar. Furthermore, because TFP, investment and government expenditure disturbances affect both the gaps and potentials, the two components will be correlated.



Figure 1: Log spectral densities of gaps and potentials: Basic SW.

To illustrate these facts, I plot in figure 1, the log spectral density of the gaps and the potentials for eight variables of the model, using one realization of the shocks, when T=750. Note that the low frequency variability of the gaps (32-80 quarters) is generally larger than their business cycle variability (8-32 quarters); that gaps and potentials almost equally account for the variance of the observables at low and business cycle frequencies; and that the correlation between gaps and potentials is, on average across variables, equal to 0.86.

To some reader, the baseline specification may appear to be unrealistic: common wisdom suggests that certain real variables should display an upward trend. To insure that this is the case, I alternatively consider a setup which allows TFP disturbances to feature a unit root. In this case, the levels of output (Y), consumption (C), investment (I) and real wages (W) will be on a balanced growth path. Because TFP drives both the gaps and the potentials of these four variables, both latent components will display an unbounded peak at frequency zero (produced by the unit root) and will respond to transitory disturbances. Thus, as in the baseline specification, the gaps for these four variables will still feature significant low frequency variability; gaps and potentials will display similar spectral shape, and will be correlated. For the variables not affected by the unit root (hours (H), return to capital (RK), capacity utilization (CapU), inflation  $\pi$ ), the baseline scenario applies. It is also worth examining what happens to the spectral shape of the latent variables when, as in Aguiar and Gopinath [2007], "the trend is the cycle", i.e. the permanent TFP disturbance accounts for a large portion of the data variance at business cycle frequencies. In principle, none of the features I discussed should be altered, but the relative variability of the gaps at low and business cycle frequencies may be affected. To generate this setup, I decrease by 3/4 the persistence of the transitory disturbances.



Figure 2: Spectral properties of gaps: Basic SW, SW with unit roots, SW 'trend is the cycle'.

Figure 2 which plots the gaps in the baseline, in the unit root and in the "trend is the cycle" setups confirms that the qualitative features of the log spectra of the gaps I emphasize remain unchanged. In particular, even in the last scenario, gaps will still display important low frequency variability.

Although the DGP I employ is popular, it does not account for financial frictions, nor it takes into account the relationship between the real and the financial side of the economy. Given the interest that the macro-financial links have generated since the financial crisis, one may be curious as to whether the presence of finance considerations and financial frictions alter the spectral properties of the gaps. In the models with financial frictions, I define the potentials in the same way I have done without them: they characterize the economy without nominal rigidities and with the markups, risk premium and inflation target disturbances set to zero. Thus, financial frictions (and risk shocks if present) affect the potential economy and the gaps.

Figure 3 compares the spectral properties of the gaps produced in the baseline specification (SW); in the baseline specification with financial frictions (SWFF), see Del Negro, Giannoni, and Schorfheide [2015]; and in the baseline specification with risky contracts (CMR), see Christiano, Motto, and Rostagno [2014]. The features of the gaps I emphasized are independent of the existence of financial frictions, except for the return to capital, which displays much larger business cycle and high frequency variations when financial frictions matter. If anything, the relative importance of low frequency components in the gaps grows larger. Quantitatively, the CMR model displays lower log



Figure 3: Spectral properties of gaps: Basic SW, SWFF and CMR.

spectral densities because the disturbances have smaller variability. Although not reported to save space, potentials either have unchanged spectral features (SWFF), or display magnified variance in very low frequency portion of the spectrum (CMR).

It is also useful to point out that policy models built according to different principles also produce gaps with the features I discuss. As an illustration, figure 4 plots the log spectra of the output gap and of the output potential produced by one simulation of ECB-Base, the Euro are version of US-FRB model. Clearly, there is an important portion of the gap variance in the low frequencies: over 40 percent of the total variance of the process is in this area; and it contributes to about 80 percent of the total output variance at those frequencies. Two features in this kind of models are however different: the potential (which is generated with a production function approach) is basically a unit root process and it contributes much less to the output variance away from the zero frequency; output gap and potential output are driven by independent disturbances.

One final exercise may provide useful information on the generality of the experiment. In the baseline specification, TFP, investment and government spending disturbances drive both potentials and the gaps. Given that these shocks are highly persistent, the conclusions that potentials and gaps have similar spectral features follows as a corollary. What happens if investment and government spending disturbances are absent? Would that make gaps and potentials different, given that markup, risk premium, and monetary policy disturbances affect only gaps? Figure 5 shows that this is not the case and none of the features I discuss is altered by this change <sup>1</sup>.

In sum, in the class of models I consider, the persistence of the disturbances (rather than their

<sup>&</sup>lt;sup>1</sup>I have also generated gaps and potentials in the model of Beaudry et al. [2018], where there is a TFP and preference shock (broadly interpreted as demand shock and interchangeable with a monetary policy shock). With an appropriately parameterization, the model is able to generate a peak the spectral density at low frequencies of the hours. However, because the endogenous mechanism generating this peak affects both gaps and potentials, gaps will display important low frequency components, gaps and potentials will have similar spectral shapes, and will be correlated.



Figure 4: Spectral properties of output gap and potential: ECB-Base.

nature) determines the low frequencies properties of gaps. Any model featuring persistent TFP disturbances and the definition of potential I employ will make gaps and potentials correlated, induce similar spectral distribution of variance, and produce large low frequency variability in the gaps.

What are the spectral features of the transitory components the SW model generates? Figure 6 shows that when transitory shocks have standard persistence the distribution of the variance by frequencies of gaps and the transitory components is qualitatively similar (compare "gaps" with "tra1"). However, while gaps and potentials are correlated, transitory and permanent components are uncorrelated. Figure 6 also shows that the relative importance of transitory low frequency fluctuations decreases in "the trend is the cycle" scenario (compare "tra1" and "tra2"). Thus, while this scenario comes closer to produce what certain filters assume, one should also be aware that the importance of the permanent component at business cycle frequencies is magnified relative to the baseline setup. Thus, the association between business cycle frequencies and transitory fluctuations will be poor.

I summarize the properties of output gaps (transitory components) for the DGPs I examined in table 1. All in all, there seems to be little loss of generality in taking the basic SW model as the DGP for the experiments. Furthermore, the setup with stationary but persistent disturbances provides a useful benchmark to estimate the distortions produced by standard filters for gap extraction. The scenario with a unit root in TFP, on the other hand, seems a reasonable alternative to examine the



Figure 5: Spectral properties of gaps: Basic SW with 7 or 5 disturbances.



Figure 6: Spectral properties of gaps: stationary and non-stationary SW.

performance of filters when extracting transitory fluctuations  $^{2}$ .

<sup>&</sup>lt;sup>2</sup>In policy circles, the idea that permanent disturbances can be represented with a random walk may be considered unreasonable, given that one often hears the view that "drivers in the long run are smooth". One could clearly use ARIMA processes for the disturbances, but this will not affect any of the features this section emphasize. The results obtained with ECB-Base model, which uses a production function approach to generate potentials confirms this. Also

	All	Low	BC	Own variance	Own variance
	frequencies	frequencies	frequencies	low frequencies	BC frequencies
Gap (SW stationary)	0.58	0.66	0.50	0.16	0.07
Gap (SW $5 \text{ shocks}$ )	0.64	0.65	0.91	0.10	0.04
Gap (SW-FF)	1.06	0.91	1.01	0.22	0.05
Gap (CMR)	< 0.01	< 0.01	< 0.01	0.15	0.02
Gap (ECB-Base)	0.01	0.35	0.27	0.33	0.40
Transitory (SW unitoot)	0.01	0.80	0.64	0.18	0.10
Transitory (SW trendiscycle)	< 0.01	0.61	0.36	0.22	0.44

Table 1: Relative variances

Notes: SW is the standard model, SW-FF is the standard model with financial frictions; SW 5 shocks is the standard model without investment and government spending shocks. CMR is the model of Christiano et al. [2014]. Base is the Euro area version of the US-FRB model. The first three columns present the faction of the variance of the observable output due to the gap, over all, low. or business cycle frequencies. The last two columns report the fraction of the variance of the output gaps (transitory output components) at low (32-80 quarters) and at business cycles (8-32 quarters) frequencies. Sample T=750. Numbers may exceed 1 because latent components are correlated.

#### 4 The filtering procedures

Given time series for the endogenous variables, I will be interested in analyzing which filter most accurately characterizes the gaps (or the transitory components) the model generates. There are numerous procedures a researcher can employ to extract two latent components from one observable variable. I focus on the most commonly used in the macroeconometric literature. I do not consider production function based procedures, because estimates of the long run values of inputs need demographics, participation rates, and other slow moving variables that can not be produced within my framework of analysis. The procedures I consider differ in many dimensions. Some are statistical and others have economic justification; some are univariate and other multivariate; some require parameter estimation and others do not. From my point of view, the two most important differences are the assumed properties of the trend; and the correlation between unobservables.

The first approach is the oldest and maintains that the trend is deterministic and uncorrelated with the cycle. Thus, the latter can obtained as the residual of a regression of the variable on a polynomial trend. I use a quadratic polynomial and run the regressions variable by variable, meaning that I do not exploit the fact that the trend will have common features in my dataset. The results obtained with this apporach are denoted by the acronym POLY in the text and the tables.

The second approach is the Hodrick and Prescott filter. Here the trend is assumed to be stochastic but smooth and uncorrelated with the cycle. The latter is the difference between the level of the series and the Hodrick and Prescott trend, which is obtained via the ridge estimator:

$$\tilde{y} = (H'H + \lambda Q'Q)^{-1}H'y \tag{1}$$

with ARIMA disturbances, potentials and gaps will still be correlated, will display similar frequency distribution of variance, and gaps will feature important low frequency variations. The use of ARIMA disturbances, on the other hand, complicates the extraction of transitory components because an important portion of the variability of the data at all frequencies is due to the permanent component.

where  $\lambda$  is a smoothing parameter,  $y = (y_1, \dots, y_t)$  the individual series,  $\tilde{y} = (\tilde{y}_1, \dots, \tilde{y}_t, \tilde{y}_{t+1}, \tilde{y}_{t+2})$ , the

denote the results with the acronym HP. In the sensitivity analysis, I consider  $\lambda = 51200$  (acronym: HPa), a value close to the BIS recommendations, see Borio [2012]. With this value of  $\lambda$ , the trend has only minor variations and the cycle accounts for the majority of the fluctuations in the data.

The third approach assumes that the trend is stochastic, displays at least one unit root, and it is uncorrelated with the cycle. I consider two separate sub-cases: one where the cycle is obtained by short differencing (one quarter); and one where it is obtained by a long differencing (24 quarters) the observable variable. I denote the results in these two sub-cases with the acronyms FOD and LD. In sensitivity analysis, I also consider 4 and 16 quarters differencing operators (acronyms: FODa, LDa)

The fourth approach permits the trend to be stochastic and to display both stationary or nonstationary features, and assumes that its variability is entirely located in the low frequencies of the spectrum. To extract cycles with 8-32 quarters periodicity, I use the band pass filter implementation of Christiano and Fitzgerald [2003], which employs asymmetric and non-stationary (time-dependent) weights. I denote the results with the acronym BP. For sensitivity, I also examine the trigonometric version of the filter see Corbae, Ouliaris, and Phillips [2002] (acronym:Trigo), the stationary, truncated symmetric version suggested by Baxter and King [1999] (acronym: BK), and a version of the baseline filter, which extracts cycles with periodicities of 8-64 quarters (acronym: BPa)

As alternative, I have also considered the wavelet filtering approach, recently suggested by Lubik et al. [2019]. Wavelet are one-sided MA filters, where the length of the MA polynomial depends on the cycles being extracted. For example, to exact cycles with a 8-32 quarters periodicity, a MA(16) is used but to extract cycles with a 32-64 quarters periodicity, a MA(32) is employed. Wavelets have some intuitive advantages over band pass filters as they work in time domain and the number of MA terms is finite. I denote the results obtained with this approach with the acronym Wa.

The fifth approach is based on local projections and follows Hamilton [2018]. Here the trend is defined as the medium term predictable component of a variable and it is obtained by running a regression of each variable at t+m on current and up to d lags of the variable. The cyclical component is assumed to be uncorrelated with the trend and obtained as the residual of the regression. I set m=8, d=4, and report results with this approach with the acronym Ham. In the sensitivity analysis, I consider the alternative of m=12 and d=2 (acronym: Hama)

The sixth approach is based on a state space formulation of the latent variable problem. It assumes that the trend is a random walk with drift; that the cycle is an AR(2) process, and allows the innovations in the trend and cycle to be correlated. No measurement error is included in the measurement equation. Using a flat prior on the parameters, I compute posterior distributions using a MCMC approach, as in Grant and Chan [2017], using 30000 burn-in draws and saving 5000 draws. The reported properties are computed averaging the resulting trends and cycles estimates over retained draws. I denote the results with the acronym UC. In the sensitivity analysis I also consider a bivariate UC filter with output and capacity utilizations as observables (acronym UCbiv).

The approaches so far described use univariate information to separate the two latent components. Given the general equilibrium nature of the DGP, these approaches are inefficient as they disregard, for example, the presence of balance growth, when there is a unit root in TFP, or the fact that cyclical components have similar features (since they are driven by the same disturbances). The next two procedures account for the possibility that commonalities may be present. The first, based on Beveridge and Nelson [1981]' decomposition, defines the trend as the predictable long run component of a vector of variables. The cycle is the difference between the vector of variables and the estimated trend. Here trend and cycles are driven by the same shocks (and thus perfectly correlated), which are the reduced form innovations of a vector autoregression on lags of the relevant variables. I run the decomposition unrestricted, that is, without the signal-to-noise prior restriction of Kamber, Morley, and Wong [2018], because the DGP is already a very low order VAR(p) polynomial.

The second procedure follows Blanchard and Quah [1989] and still uses a vector autoregression to compute the shocks to the variables. However, rather than the vector of innovations, it uses identified disturbances to separate the two latent components. Here, the trend is driven only by supply disturbances and the cycle by both supply and demand disturbances. In the implementation I use, the vector autoregression includes output growth and hours for both approaches. I denote the results with the acronyms BN and BQ. In the sensitivity analysis, I also consider a trivariate VARs with output growth, consumption to output, and investment to output ratios (acronyms: BNa, BQa). While there are other multivariate approaches one considered, the list of procedures I employ is sufficiently exhaustive and covers well what is available in the literature, thus making the comparison exercise meaningful, and the results informative.

#### 4.1 What should we expect?

Given that different methods use different assumptions to identify the latent components, one should expect them to produce different outcomes, see Canova [1998]. Furthermore, since no procedure takes into account the features of the DGP I discussed in the previous section, biases might be expected.

To be specific, procedures which assume no correlation between the two latent components, should be relatively poor when extracting gaps but better endowed when extracting transitory fluctuations. A deterministic polynomial approach is likely to overestimate the volatility of both the gaps and of transitory fluctuations, given that both potentials and permanent components are stochastic; while methods which impose a unit root should be better suited to separate permanent and transitory components than potentials and gaps. Finally, methods designed to focus attention on cycles with particular periodicity, will distort the frequencies distribution of the variance of gaps and of transitory fluctuations. In fact, they will attribute the low frequency variations belonging to both components to the trend, and the cyclical (and high frequency) fluctuations belonging to both components to the cycle. Because model-based gaps and transitory components display variability at all frequencies, while such approaches carve the spectrum by frequencies, biases will be significant, see Hansen and Sargent [1979] for an earlier statement of this problem, and Canova [2014] for a recent one.

The presence of unit roots is unlikely to affect much conclusions obtained without and distortions will be generally present also in this case. Misspecification of the properties of the DGP is the reason for their existence. For example, since a number of procedures assume that the permanent components are random walks, while the growth of permanent components the model generates are highly serially correlated, estimates obtained with these methods will distort the properties of the transitory components. Perhaps more importantly, many approaches will twist the frequency distribution of transitory variability. Thus, turning point dating, measures of durations and amplitudes of expansions and recessions and cross-variable relationships will be generally distorted. Hence, no procedure is close to the ideal given the DGP I consider. Moreover, because different procedures employ different assumptions to identify the latent variables, they will produce different biases. The question of interest is which one is least damaging and why. Note that because the distortions I discuss hold in population, small samples can add to the problems, especially for those procedures which require parameter estimation. Finally, while tinkering with the filters may lead to some quantitative improvements, it is unlikely that any modification will dramatically change the fact that no procedure is close to the ideal.

#### 5 The statistics

To measure the performance of different approaches, I consider a number of standard statistics. These are computed averaging results using 100 data replications, to wash out simulation uncertainty. First, I compute the mean square error (MSE), calculated as the difference between the true gaps (the true transitory components) and the filtered series.

Second, I report the contemporaneous correlation between the true gaps (true transitory components) and the filtered series, the first order autocorrelation and the variability of the filtered series, benchmarking them with those of the true gaps (the true transitory components). I compute these four statistics for 9 series that the model generates: output, consumption, investment, return to capital, hours, real wages, capacity utilization, inflation and nominal rate. Note that the latter two series have no potential. I also extract a factor from the actual data, apply the filtering procedures, and compare MSE, variability, auto and contemporaneous correlations of the filtered series to the those of the factor computed using the true gaps (true transitory components).

Third, I compute turning points in the filtered data and compare their number, the average duration and the average amplitude of expansions and recessions with those present in the true gaps (the true transitory components). Because in the baseline case, the model is solved linearly around the steady state, duration and amplitudes are roughly symmetric across business cycle phases in the simulated gaps (transitory data). This may not necessarily be the case in the filtered series

Policymakers are interested in latent variables for two reasons. First, because the want to measure in real time the state of the economy. Second, because they want to use them to predict other variables, for example, inflation via a Phillips curve, or employment (hours) with a Okun law. For this reason, I also compute two additional set of statistics. The first measures the MSE in real time, focusing attention on the last 12 periods of each sample; the second compares the variance of the prediction error in the regressions implied by the true gap (transitory) data and those implied by the filtered data. Letting  $y_{t-j}^i$  be either the true output gap (true transitory output) or the filtered one and m=1,4, the predictive regressions take the form:

$$\pi_{t+m} = \alpha_0 + \alpha_1 \pi_t + \sum_{j=1}^3 \beta_j y_{t-j}^i + e_{t+m}$$
(2)

$$h_{t+m} = \alpha_0 + \alpha_1 h_t + \sum_{j=1}^3 \beta_j y_{t-j}^i + e_{t+m}$$
(3)

#### 6 The results

Table 2 reports a summary of the results counting, for each statistics, the number of times a procedure is least distorting across variables. Counting measures assign a one to the best procedure (0.5 if there is a tie) and zero to the others. Totals are computed equally weighting all statistics. Tables 4-10 in the appendix give the details: for each variable (factor) and for each procedure, table 4 reports the average MSEs across replications; table 5 the average real time MSEs; table 6 the average contemporaneous correlation, table 7 the average AR1 coefficient; and table 8 the average variability; table 9 the average number of turning points, the average durations and amplitudes of recessions and expansions for output and the factor; and table 10 the average variance difference in the prediction error of the Phillips curve and Okun law regressions between each procedure and the true one.

Statistic	POLY	HP	FOD	LD	ΒP	Wa	Ham	UC	BN	BQ	BW
				(	Gap	)S					
MSE	5	3						1	0.5	0.5	8
Corr	9								0.5	0.5	8
AR1	4			3		3					6
Var	4			2		3	1				
TP	1.5	5	2	1.5							3
RT-MSE		1				3	2	3	0.5	0.5	8
PC	2										2
OL				1			1				
Total	25.5	9	2	8.5	0	9	4	4	1.5	1.5	35
				Tra	nsi	tory					
MSE			9					1			
Corr											
AR1	4					5.5		0.5			1
Var	3			6			1				1
TP	4	4		2							3
RT-MSE			4					6			
PC				1						1	
OL				2							
Total	11	4	13	11	0	5.5	1	7.5	0	1	5

Table 2:	Summary	results,	T = 750
----------	---------	----------	---------

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Wa a wavelet filter, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. MSE is the mean square error, Corr the contemporaneous correlation with the true series, AR1 the first autoregressive coefficient, Var the variability of the series, TP the number of turning points, the duration and the amplitude of expansions and recessions, RT-MSE is the real time MSE, PC is the Philipcs curve prediction, OL the Okun law prediction. In each row the ranking is over 9 series and one factor, except for TP where the ranking is for output and the factor. Numbers are computed summing the top ranks, equally weighting all variables; ties each get value of 0.5. Each table has two parts. In the upper panel, I report the performance of different filters for gap extraction; in the second their ability to recover the transitory component of the data.

The Polynomial approach is, by far, the least distorting among the competitors when measuring gaps. The approach is superior as far MSE, contemporaneous correlation, AR1, variability, and Phillips curve regressions are concerned, while it lags behind in terms of real time MSE, turning points characterization, and Okun law regressions. The next approaches in the ranks are appropriate for some statistics only (the HP filter for TP detection; the Wavelet filter for TP detection, AR1 coefficient and variability) but seem to be much less suited than the Polynomial filter to capture features of the gap data. Three additional aspects of the top panel of table 2 are worth emphasizing. The long difference filter is superior to the Hamilton filter, suggesting that the horizon of the local projection is probably too short: Hamilton [2018] shows that the filter is close to a eight-period difference filter. The commonly used UC approach, on the other hand, is competitive only in terms of real time MSE; for other statistics, it is never among the top procedures. Finally, the performance of the bivariate BQ and BQ procedures is poor: they top other approaches for hours in terms of MSE, real time MSE, and contemporaneous correlation, but never rank first for output gaps.

Quantitatively speaking, the distortions in terms of time paths are large and biases relative to theory-based measures of gaps important. For example, the average MSEs are all larger than the average MSE produced by a random walk (which, e.g., for the output gap is 14.45), and the real time MSE are almost twice as large as the real time MSE produced by a forecast that uses T - 12 value for all successive 12 periods. The magnitude of the distortions is also large in terms of volatility and for correlation measures; in the latter case, except for the Polynomial procedure. At the opposite extreme, biases in the persistence parameter are relatively small and, except for the FOD filter, all estimates are in close range of the true AR1 coefficient.

When it comes to characterizing the properties of transitory components, the FOD filter is the least distorting, closely followed by the Polynomial and LD filters. However, the superiority of the FOD filter is entirely concentrated in MSE measures, while the Polynomial and LD approaches come on top for a number of statistics. In general, the distribution of winners is more uniformly spread across procedures than with gap measures. Relative to gap extraction, one can notice a deterioration of the performance of HP and Hamilton filters and an improvement in the performance of the UC filter, which entirely comes from the real time MSEs, and involves output, consumption and investment. That is to say, the filter seems appropriate as far as predicting in real time the transitory component of output, consumption and investment up to 12 quarters ahead, but gives a poor characterization of their historical properties. BN and BQ performance is poor, despite the fact that they use more information to extract the latent transitory variables. While the inferiority of BN and BQ could have been expected in the case of gaps because they overdifference output data (the output series is stationary), it is less obvious why this is the case when extracting transitory components. Given the popularity of BQ procedures in the literature, the next section studies why this is the case. Finally, notice that the band pass filter is never the best for any statistics or series for extracting both gaps and transitory components and that the Wavelet filter does a reasonable job in comparison. Because both filters aim at capturing portions of the spectrum, the differential performance must due to the smaller approximation error of wavelet filter possesses.

The magnitude of the distortions when extracting transitory components is generally larger than when extracting gaps. As we have seen, the presence of a unit root changes the relative distribution of variances of the permanent and of the transitory components in the low frequencies (the former gets more weight) and this affects the performance of a number of approaches. Interestingly, almost all procedures have a hard time to capture the ups and downs of the transitory components, see figure 7 for examples. Thus, MSE errors tend to be large and the contemporaneous correlation between the true and filtered series is essentially zero for several filters.



Figure 7: Time series: true and estimated transitory components. Various filters.

Many approaches have the tendency to underestimate the variability of the both the gaps and the transitory components. The exceptions are the Polynomial, the BN and BQ procedures. For the polynomial filter, this is expected given the nature of the squared gain function I discuss in section 7. Despite this obvious bias, it is remarkable that the procedure is still on the top for gaps and the second to the top for transitory components. The excess variability produced by BN and BQ procedures is less expected and, as shown in the next section, due to the fact that the approaches considerably distort the frequency distribution of the variability of the latent series and erroneously attribute low frequency variations present in the gap (transitory) data to the trend component.

To summarize, and somewhat unexpectedly, the Polynomial approach, the crudest and the oldest method existing in the literature, turns out to be the least damaging in characterizing the properties of both the gaps and the transitory fluctuations. A similarly crude first order (long) differencing filter ranks well when it comes to extract the transitory components of the data. Furthermore, using larger information sets or fancier econometrics does not seem to help.

Table 2 considers 9 series generated by the model and the factor characterizing their common dynamics. Because policymakers are mostly concerned with output gaps and output dynamics, is it worth zooming in the table and look at how different procedures characterize the latent output quantities. The HP filter is uniformly superior when measuring the number of turning points in the gap data, the duration and the amplitude of expansions and recessions. The result is striking, given the criticisms raised in the literature against the HP filter; see Hamilton [2018] for the latest installment. Next in the list are the Polynomial and the UC approaches. The next section shows why they do well for output gaps and, for the UC filter, why is not necessarily so for other series. For transitory output, the Polynomial approach ranks first, and HP and FOD come close seconds. The FOD approach tops other procedures in terms of MSE measures, while the Polynomial and the HP approches are superior when measuring the number of turning points and the durations and amplitudes of expansions and recessions. Given that the Polynomial approach has also been discredited for leaving near non-stationary dynamics in filtered output - and referees often harp about filtering data with such a procedure - these results are surprising and deserve further discussion.

#### 6.1 Sensitivity

I have repeated the Monte Carlo exercise varying the parameters of different procedures, tailoring certain filters to better capture the low frequency component of the data, or adding information to multivariate approaches. The performance of HP and BP filters can be generally improved by choosing a higher  $\lambda$  or lowering the limit of the frequency band (see tables A.1-A.7 in the on-line appendix) but, by and large, the conclusions I have presented hold.

In particular, it is still the case that the Polynomial procedure is least distorting when measuring gaps and none of the refinements is able to produce MSEs which are uniformly smaller, both for the gaps and the transitory components. The alternative HP and Hamilton filters produce estimated cycles whose frequency is more in line with those of gaps (transitory data) but this does not change the relative position of the filters in the ranks since the improvement is larger for those variables or statistics where the filters were the top approaches. The most significant changes are produced with the bivariate UC approach when computing the real time MSE; with the alternative BP filter when measuring the AR1 coefficients in gaps; and with the alternative Hamilton and LD filters when computing variabilies. Quantitatively speaking, the gains are small.

Perhaps more interesting are the results obtained with a sample of 150 observations. Table 2 used a large sample (T=750). In practice, estimation uncertainty matters, and given that some procedures require parameter estimation and others do not, some of the conclusions may be affected. A summary of the results for T=150 is in table 3 (tables A.8-A.14 in the on-line appendix provide the details). Overall, the Polynomial approach is still the best when measuring gaps, followed by HP and LD filters. Relatively speaking, a short sample worsens the performance of the Polynomial filter and improves the one of filters which do not require parameter estimation. Also, while there are statistics specific changes in the ranking, the snapshot of table 2 is broadly maintained.

When measuring the transitory components results are somewhat affected. The FOD filter, which was best in table 2, now looses its superiority and the LD filter becomes the least distorting procedure, and the UC approach lags third. The LD filter dominates when measuring the persistence and volatility of the transitory component, while the FOD filter still maintains its superiority in terms of MSE measures. The UC filter, on the other hand, does relatively well only in terms MSE measures. For the two regressions policymakers care about, the sample size makes little difference, and the LD filter is overall the most appropriate method to construct regressors in the predictions equations.

I have also conducted an additional experiment where the model is simulated using a second order solution. In this case procedures which assume a linear, parametric structure are penalized relative to procedures which non-parametrically split the data. In addition, the magnitude of the distortions is, on the whole, larger. However, the ranking among procedures is only minorly affected.

Statistic	POLY	ΗP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
				(	Gaps	3					
MSE	4	4						1	0.5	0.5	7
Corr	7						2		0.5	0.5	10
AR1	3			4.5		2.5					4
Var	5	1		3					0.5	0.5	2
TP	0.5	3.5	2	1		2	1				2
RT-MSE		3				4	2		0.5	0.5	8
PC								2			2
OL				2							
Total	19.5	11.5	2	10.5	0	8.5	6	4	1.5	1.5	33
				Tra	nsit	ory					
MSE			4		1			5			
Corr											
AR1	0.5			7		1.5			0.5	0.5	3
Var	3			7							7
TP	2	2		3	1.5			1.5			
RT-MSE			5		3			2			
PC				2							
OL			1	0		1					2
Total	5.5	2	10	19	5.5	2.5	0	8.5	0.5	0.5	13

Table 3: Summary results, T=150

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Wa a wavelet filter, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. MSE is the mean square error, Corr the contemporanceus correlation with the true series, AR1 the first autoregressive coefficient, Var the variability of the series, TP the number of turning points, the duration and the amplitude of expansions and recessions, RT-MSE is the real time MSE, PC is the Philips curve prediction, OL the Okun law prediction. In each row the ranking is over 9 series and one factor, except for TP where the ranking is for output and the factor. Numbers are computed summing the top ranks, equally weighting all variables; ties each get value of 0.5.

#### 7 GAIN FUNCTIONS

To understand the pattern of results I obtain, I first examine the estimated squared gain function of each filter. As already mentioned, this function measures the reduction/amplification at each frequency of the variance of the observable series due to the filter. Since the theoretical squared gain function for output is different from zero, bounded above and, for gaps, close to half at all frequencies (see table 1), deviations of estimated squared gain functions from the ideal shape give a synthetic idea of the effects of each filter.

Figure 8 presents the estimated squared gain functions for 12 filters (Polynomial, HP, BP, Hamilton, FOD, LD, BN, BQ, UC, Wave, HPa and BPa) using one realization of the output series when

all disturbances are stationary. Given the DGP, the spectral properties of output are similar across replications and only minor changes in the high frequency portion of the spectrum are visible. Thus, there is no loss of generality focusing on one realization. Furthermore, because the disturbances driving output and their propagation features are the same as those driving, e.g., consumption, investment or real wages, the output squared gain functions is sufficient to understand the qualitative distortions each filter produces.



Figure 8: Estimated squared gain function for output: selected filters.

Figure 8 displays some known features and some less known ones. The squared gain of the polynomial approach is zero at the zero frequency and one everywhere else. Thus, the procedure removes very long run variability and leaves the rest of the spectrum of the original series untouched. In other words, the cycles the procedure generates have roughly the same features as in the original series and the frequency distribution of the variance is similar. The squared gain of the HP filter, on the other hand, shows the characteristics high-pass features of the approach. The filter eliminates low frequency variability, keeps the high frequency variability unchanged and, at business cycle frequencies, smoothly eliminates power, when moving from cycles of 2 to 8 years. The band pass filter knocks out low and high frequency variability and passes the business cycle frequencies almost unchanged. Because the sample is finite, the filter displays some compression, in the sense that at some of the selected frequencies, the squared gain is less than one.

Perhaps more interesting is the squared gain of the Hamilton filter, which has not been yet explicitly described in the literature. Because the procedure uses  $y_{t+m}$  as dependent variable in the projection equation, the gain function is zero at m/2 separate frequencies. Between these frequencies the filter has a bell- shaped squared gain and, at the vertex, the height exceeds 10. Thus, while at certain frequencies the variability of the original series is eliminated, at others, the variability of the filtered series is 10 or more times larger than the variability of the original series. Apart from this severe alteration in the distribution of the variance by frequency, it is clear that the gain function does not resemble the one of a business cycle filter: it emphasizes frequencies of the spectrum not necessarily connected with meaningful cycles and creates excess variability at 'uninteresting' frequencies. Hamilton [2018] mentions that the cycles his approach produces are similar to those generated by a LD filter. Figure 8 confirms that the squared gain of the latter has, qualitatively, the same features as the Hamilton filter. Nevertheless, because I take a 24-quarter rather than an 8-quarter difference, the LD filter also emphasizes low frequency variability.

The FOD approach has a familiar squared gain function: it attributes all the variability of the original series in the low and business cycle frequencies to the trend while the cycle captures, primarily, very high frequency variability.

The BN and the BQ filters have qualitatively similar squared gain functions, despite the fact that they identify latent components with different assumptions (in the BN decomposition the two components are correlated; in the BQ they are not). Quantitatively, this source of misspecification matters for the results and the BN filter is, in general, worse. Because the squared gain function at the very low frequencies is large for both filters, the estimated cycles display strong very low frequency variability and considerable persistence.



Figure 9: Squared Gain: UC filter

The UC and the Wavelet filters have gains functions that are different from the others. In particular, the squared gain of the UC filter is less than one at all frequencies and has a vertex of around 0.4 in the low frequencies. Thus, it seems to recognize that gaps display variability at all frequencies and possess substantial power in the low frequencies. However, this shape of the gain function is specific to the output series. As shown in figure 9, the squared gain for, e.g. consumption,

produces a cyclical component with a large magnification in the variance at business cycle and at certain high frequencies. The Wavelet filter also has a squared gain function with no zeros except at the zero frequency. However, it magnifies the variability of the original series at low frequencies and this accounts for its somewhat mixed performance in the exercises I run.

Figure 8 also plots the gain function of the alternative HP and BP filters, both of which are designed to capture cycles from 8 to 64 quarters periodicity. Clearly, by changing the parameters of the filters, one can shifts the low frequency variations present in the observable data to the cycle. Still, these changes do not reduce the squared gain at business cycle or high frequencies. Thus, all the variability at these frequencies is still mistakenly attributed to the cycle and this accounts for the generally inferior performance of the alternative HP and BP approaches in the overall rankings.

Although the quantitative details change when the TFP disturbance has a unit root, the estimated squared gain functions are roughly similar (see figure 15 in the appendix). Noticeable differences occur for the BP filter, whose compression increases; the Wavelet filter, whose magnification in the low frequencies increases; and for BN and BQ filters, whose estimated squared gain function changes shape and it is now qualitatively similar to the one of the FOD filter. Thus, apart from the BN and BQ filter, none of the procedures sees any difference in the two exercises I run and fail to recognize that gaps and transitory components are objects with different statistical properties and different economic interpretation.

#### 7.1 DISCUSSION

The evidence I produced indicates that the Polynomial approach is superior to other filters when extracting the gaps and figure 8 explains why. Because the estimated squared gain is one at the majority of the frequencies, there is a general overestimation of the gaps variance but the persistence and variance share by frequency are roughly matched. Thus, the ups and downs in the filtered series and in the theory-based gaps are similar in terms of timing and durations. The same argument holds, in part, when extracting transitory components. However, because transitory components display a smaller amount of very low frequency variability, distortions are larger.

Because in the DGP potentials explain an important portion of the variance at business cycle frequencies and gaps explain a large fraction of the variance at low frequencies, HP and BP filters distort the frequency distribution of the variance of the latent variable of interest. In fact, most of the low frequency variations are attributed to the trend and all the business cycle variations are attributed to the cycle. This means that the persistence of the gap processes is underestimated and that the filtered series have patterns of ups and downs that do not generally match those of theory-based gaps in terms of timing, durations, and amplitudes. While the frequency distribution of the variance of the two latent components changes when TFP has a unit root is smaller, HP and BP filters fail to capture the fact that the low frequency transitory variability is significant. Thus, distortions are also large also in this case.

Since the HP filter is the leading procedure to extract cycles in international institutions (e.g. BIS or OECD), further discussion is warranted. The standard HP filter uses a smoothing parameter of  $\lambda = 1600$ , which typically is interpreted as indicating that the standard deviation of the cycle is 40 times larger than the standard deviation of the second difference of the trend. Hamilton [2018] criticizes this choice of  $\lambda$  suggesting that estimates of the ratio obtained in state space models that approximate a one-sided HP filter are much smaller. When I compute the range of theoretical  $\lambda$  values obtained by taking the variability of the gaps (transitory components) to the second difference

of the potential (permanent components) across series I find that indeed they are much smaller than 1600 and in the range of [3,24]. However, as already mentioned, gaps display quite a lot of low frequency variations and only when  $\lambda = 51200$  these variations become part of the estimated cycle. Note that such a  $\lambda$  value is close to the one typically used to extract financial cycles and that, with  $\lambda = 51200$ , the absolute performance of HP filter in the Monte Carlo exercise improves. Thus, when the two latent components have similar spectral properties, are potentially correlated, and the gaps are not iid,  $\lambda$  does not have the standard interpretation given in the literature. In other words, application of standard state space approaches to estimate the smoothing parameter would lead to important distortions when attributing low frequency components to trends or to cycles <sup>3</sup>



Figure 10: True output gap, true output transitory and estimated Hamilton cycles.

Hamilton [2018] suggested that the local projection filter is a reliable tool to extract cyclical fluctuations. My Monte Carlo exercise shows that when data is generated by standard models, the filter does poorly in extracting both gaps and transitory components. A few reasons may explain the outcomes. First, gaps and potentials are correlated while the projection equation used to separate them assumes that they are not. Second, even though the approach does not impose unit roots, unit roots are in fact removed (these are the zeros in the estimated squared gain function). Even when a unit root is present in the data, the approach overdifferences the data, thus creating spurious MA components in the estimated cycles. While theoretically important, these two reasons may not be crucial to explain the poor performance of the approach because the LD filter, which implicitly errs in the same direction, has a much better performance than the local projection filter. The main

<sup>&</sup>lt;sup>3</sup>The UC setup I use is not appropriate to quantify the magnitude of the distortions in  $\lambda$  because the trend is correctly assumed to be a random walk.

reasons for the poor performance of approach are: the filter does not recognizes low frequency gap variations and attributes them to the trend; and it magnifies the importance of certain high frequency variations, which have little economic interpretation. Figure 10 illustrates the difficulties of the filter in replicating the dynamics of the output gap (transitory output) in one simulation.



Figure 11: True and estimated transitory components: BN and BQ filters.

VAR-based decompositions perform well for hours but not for output and the distortions in terms of MSE, correlation with the true component, or variability are large, both in absolute and in relative terms. This is true when all shocks are stationary and when the TFP shock has a unit root. When shocks are stationary, the VAR used to extract latent variables is misspecified (since output is stationary, it is overdifferenced) this may be one of the reasons for the poor performance of BN and BQ approaches for output gap extraction.

When TFP has a unit root, the overdifferencing problem disappears. Still, both approaches are poor in extracting the transitory component of output. To illustrate the problem, figure 11 presents the estimated BN and BQ and true transitory output component for one simulation. Clearly, the estimated BN transitory component shows considerable low frequency variations which are absent from the true transitory component and due to the fact that the filter forces the transitory and the permanent components to be highly correlated. The BQ filter also produces a transitory component with low frequency variations, but also increases high frequency variations. Hence, in both cases some low frequency variations present in the theory-based permanent components are erroneously attributed to the transitory component.



Figure 12: True and estimated potential and permanent components BQ filter.

Coibon et al. [2018] have argued that a BQ decomposition can be used to measure the dynamics of potential output. My results do not support their choice. Instead they indicate that, if the data has been generated by a DSGE model, the estimated permanent component will tend to overstate the dynamics of potential fluctuations. Figure 12, which plots potential and permanent output components in one simulation together with the estimated BQ permanent components, clearly shows the problem. The BQ permanent component displays too much low frequency variations (there are very long drifts in the ups and downs) and too little medium-business cycle variations.

Why is the BQ decomposition so poor? It turns out that the VAR used to extract latent variables is misspecified, even when TFP has a unit root, because of the deformation problems studied in Canova and Ferroni [2019]. In particular, the VAR model is bivariate (trivariate) while the DGP has seven shocks; not all the states of the DGP enter the empirical model. Thus, not only seven shocks are compressed into two (three) innovations; since states are omitted, some VAR innovations are serially correlated even when a generous lag length is used. Deformation problems increase the persistence of the estimated shocks and mess around the correlation structure between the estimated and actual shocks. Indeed, the estimated BQ transitory output series are more persistent than the true transitory output series and the correlation between the TFP disturbances and the estimated supply shocks is low (0.43) since the latter captures a number of stationary demand shocks present in the DGP. When the sample is short (T=150), standard problems estimating long term quantities with small samples are added, see Erceg, Guerrieri, and Gust [2005]. Since a long lag length is needed to reduce states omission in the VAR, parameter estimation may be further compromised, making inference about the latent components problematic.

Ramey and Zubairy [2018] have used estimates of trend output to scale down the variables prior to the computation government spending multipliers. While they use a polynomial approach and thus minimize the distortions if the DGP belongs the class of DSGE models I consider, the practice is dangerous since inference depends on the quality of the preliminary trend output estimates.

#### 7.2 Optimal filters for DSGE-based gaps

None of the procedures I have employed approximates the ideal filter for the class of DGPs I consider, making inferential distortions potentially large. When the proportion of the variance of the two latent components at medium and business cycle frequencies is roughly similar, standard filters generate biases. Different filters carve the spectrum differently, but there is the tendency to attribute most of the low frequency variations to the trend and the majority of the business cycle variations to the cycle, muting the persistence and the dynamic properties of the estimated cycles, altering the sequence of turning points, and the properties of amplitudes and durations of business cycle phases.



Figure 13: Squared gain functions, Butterworth filters.

How does one then proceeds in practice? One obvious way is to setup a structural model, estimate its parameters by conventional methods, and with mode estimates and some initial conditions, generate model-based latent quantities of interest. Examples of such an approach exist in the literature, see e.g. Justiniano et al. [2013] or Furlanetto et al. [2020]. Clearly, if the sample is short and the prior insufficiently tight, estimates of the latent variables will reflect the noise present in the parameter estimates. Furthermore, if the model is misspecified, biases will litter estimates of the latent quantities. While not much can be done about small samples, model misspecification can be taken care, in part, with the approach of Canova and Matthes [2018]. The setup is particularly useful because it forces consistency of parameter estimates across models with different features and this may help to robustify the measurement of latent quantities.

While the above solution is appealing, most researchers may still prefer to be agnostic about the process generating the data and, when selecting the filter to employ, only willing to take into consideration the frequency domain features I have emphasized in section 3. In this case, is there an alternative filter whose gap estimates are uniformly superior to those I consider? Engineers extensively employ Butterworth filters in their analyses, because they are flexible, have uniform squared gain across the frequencies of interest, have a ARMA representation, and do not feature the side loops. In particular, Butterworth filters can be designed to make sure that the estimated cycle features significant low frequency variations and the estimated trend significant business cycle variations. Figure 13 shows the gain function of a number of such filters as function of the polynomial order (n) used to filter the data (reported are n=1,2,4); the cutoff point  $\omega$ , where the squared gain declines (reported are  $0.95\pi$ ,  $0.75\pi$  and  $0.50\pi$ ); and of the scale parameter  $G_0$ , determining the height of the squared gain (reported are  $G_0 = 1, 0.4$ ). Clearly, by appropriately choosing the three parameters, one can give the function a variety of interesting shapes and mimic, e.g., low pass, high pass, or band pass filters. In fact, it can be shown the HP and BP filters I have used in this paper, are special Butterworth filters. For the purpose of gap estimation, the most relevant gain function is the black one (with circles). Note that has a uniform height of 0.4 up to  $\omega = 2/3\pi$ . Thus, it would attributed a portion of the low frequency variance of the data to the cycle and a portion of the business cycle frequency variance to the trend.



Figure 14: Output gap and Butterworth filtered series and log spectra.

Figure 14 reports the output gap generated by one simulation together with the estimated cycles obtained with different Butterworth filters and their log spectral density. The filtered series display the low frequency movements present in the true output gap and replicate quite well the distribution of variance of the process in the low and business cycle frequencies. When compared with the time paths reported, e.g. in figures 10 or 11, the match is clearly superior. The results are not specific to the output gap series. Figures A1-A4 in the on-line appendix present detailed information for 8 gaps series and 8 transitory series generated by one simulation of the model, using the same Butterworth filter for all series (for the gaps, a low pass filter,  $n=1, \omega = 0.95\pi$ ,  $G_0 = 0.4$ ; for the transitory components, a high pass filter,  $n=1, \omega = 0.004$  and  $G_0 = 1$ ).

The last column in tables 2 and 3 report the number of times Butterworth estimates are superior to those obtained with the best performing method for each statistics across series and simulations; the last column of tables 4-10 gives the average statistics across Monte Carlo simulation. It is remarkable that the Butterworth cycle is closer to the actual gap series than the previously selected best approach in 37 out of 66 cases when T=750 and in 33 out of 66 cases when T=150, and improves the average contemporaneous correlation with the actual gap series for all variables and both sample sizes. Thus, if one prefers a statistical to a model-based filter when measuring gaps, a careful design of Butterworth filters may help to solidify economic inference.

The performance when extracting the transitory component is less impressive, as it is difficult to twist a Butterworth filter to produce a cycle which simultaneously displays larger variability at low frequencies and smaller variability at business cycle frequencies relative to the permanent component (as reported in table 1). To do so, one would need a non-uniform gain function across the frequencies of interest Still, even in this less than ideal case, the approach does better or, at least, not worse than the previously selected best approach in over 1/3 of the cases the tables consider.

#### 8 CONCLUSIONS

The punchline of the paper is simple. If the data has been generated by the class of models macroeconomists employ to interpret the dynamics of aggregated data and to provide out-of-sample predictions, then the toolkit of filters available to separate the observable data into latent components is inappropriate. If one has to choose among the lame horses that are available, the oldest (Polynomial) and the simplest (differencing) turn out to be the best.

These conclusions obtain because in theory, gaps and transitory fluctuations both have substantial low frequency variability; potentials and permanent fluctuations both have considerable business cycle variability, and the frequency distribution of the variance of gaps and potentials is similar. In this situation, filters that carve the spectrum by frequencies are unsuitable, because 'the trend is the cycle and the cycle is the trend', as are methods that fail to recognize the nature of low frequency variations, and that emphasize either high frequency variations, or fluctuations located at uninteresting frequencies. The polynomial approach produces the smallest distortions when measuring the gaps because, away from the zero frequency, it leaves the frequency distribution of variance of the level data unchanged. For the same reason, it is also close to the top when one wants to recover the transitory components. On the other hand, long differencing works better for transitory fluctuations, because it is able to attribute a portion of the low frequency variations to the cycle.

Because the performance of popular filters is far from satisfactory, one should be careful in using their output to evaluate the state of the business cycle, to forecast inflation or unemployment, or to provide policy recommendations. The warning is even more important when the sample is short, the filter requires parameter estimation and inference needs real time parameter estimates.

One could clearly take the position that the class of structural models I used to simulate data has nothing to do with the real world and therefore that the exercises I run have no information about the usefulness of existing procedures. I have shown that the features I emphasize also obtain in models with additional or different frictions and, even, with different organizing principles. Thus, unless one is willing to dismiss many of the currently available macroeconomic models, one must find a different way out of the conundrum I brought to light.

I have suggested two potential solutions. One solution is to go structural and to compute gaps (transitory components), conditional on a model and the estimated parameters. While straightforward, some researchers may be reluctant to follow such an approach, given that even complex models are not the DGP of the data and apparently innocent estimation choices may impair inference, see e.g. Canova [2014]. If robustness is the main concern, the approach of Canova and Matthes [2018] can be used to provide more reliable estimates of the latent components.

The other solution is to design methods that take into account the features of the DGP. A useful approach should allow low frequency variations to be present in the estimated cycle and business cycle frequency variations to be present in the estimated trend. I have described a class of filters that can be rigged to produce estimated latent components with these features and showed that they are uniformly superior to the available methods for gaps and competitive with the best approaches for transitory components. Clearly, a more detailed study of the properties of this class of filters is needed to strengthen user confidence about the approach.

Two additional implications of the results are worth emphasizing. While it is standard to think of economic and financial cycles as being distinct, in the sense that the largest share of the variances is located at different frequencies of the spectrum, see e.g. Borio [2012], the fact that models with or without financial constraints have similar features and that gaps and transitory components have considerable power in the low frequencies in both situations suggests that two phenomena may not be distinct. Perhaps, it is the insistence of macroeconomists on focusing attention on cycles of two to eight years that has given a misleading impression of what it is in macroeconomic data. I will explore this issue in future work.

There has been an industry over the last 30 years trying to collect stylized business cycle facts, both to inform the construction of realistic models and to test them, see Angeletos et al. [2019] for a recent example. These exercises generally focus attention on variations with 8 to 32 quarters periodicity and disregard the rest. Given the results I have presented, it is perhaps desirable to switch attention from 8 to 32 quarter to low frequency cycles, or at the minimum, take into consideration the fact that most of the variance in the data is not located at the so-called business cycles frequencies, see e.g. Kulish and Pagan [2019]. Paying more attentions at the spectral properties of the data will help researchers to better understand the kind of models which are consistent with the data.

#### References

- Mark Aguiar and Gita Gopinath. Emerging market business cycles: The trend is the cycle. *Journal* of *Political Economy*, 115(1):69–102, 2007.
- Marios Angeletos, Fabrice Collard, and Harris Dellas. Business cycle anatomy. Technical report, MIT manuscript, 2019.
- Marianne Baxter and Robert King. Measuring business cycles: Approximate band-pass filters for economic time series. *Review of Economics and Statistics*, 81:575–593, 1999.
- Paul Beaudry, Dana Galizia, and Frank Portier. Putting the cycle back into business cycle analysis. Technical report, British Columbia, UCL and Carltron, 2018.
- Stephen Beveridge and Charles R. Nelson. A new approach to decomposition of economic time series into permanent and transitory components with particular attention to measurement of the business cycle. *Journal of Monetary Economics*, 7:151–174, 1981.
- Olivier Jean Blanchard and Danny Quah. The dynamic effects of aggregate demand and supply disturbances. *American Economic Review*, 79:655–73, 1989.
- Claudio Borio. The financial cycle and macroeconomics: what have we learned? Technical report, BIS working paper 2012, 2012.
- Fabio Canova. Detrending and business cycle facts. *Journal of Monetary Economics*, 41:475–540, 1998.
- Fabio Canova. Reference cycle and turning points: A sensitivity analysis to detrending and dating rules. *Economic Journal*, 109:126–150, 1999.
- Fabio Canova. Methods for Applied Macroeconomic Research. Princeton University Press, 2007.
- Fabio Canova. Bridging dsge models and the raw data. *Journal of Monetary Economics*, 67:1–15, 2014.
- Fabio Canova and Filippo Ferroni. Mind the gap! stylized dynamic facts and structural models. Technical report, CEPR working paper 13948, 2019.
- Fabio Canova and Christian Matthes. Dealing with misspecifaction in structural macroeconometric models. Technical report, Norwegian Business School, 2018.
- Lawrence Christiano and Terry J. Fitzgerald. The band pass filter. *International Economic Review*, 44:435–465, 2003.
- Lawrence Christiano, Roberto Motto, and Massimo Rostagno. Risk shocks. American Economic Review, 104:27–65, 2014.
- Olivier Coibon, Yuriy Gorodnichenko, and Mauricio Ulate. The cyclical sensitivity in estimates of potential output. *Brookings Papers on Economic Activity*, Fall, 2018.

- Dean Corbae, Sam Ouliaris, and C.B. Phillips, Peter. Band spectral regression with trending data. *Econometrica*, 70(3):1067–1109, 2002.
- Olga Croitorov, Atanas Hristov, Kieran McMorrow, Phillipp Pfeiffer, Werner Roeger, and Valerie Vandermuellen. The eu cam and the relationship between output gaps, cyclical indicators, and inflation. Technical report, European Commission, DG ECFIN report, 2019.
- Stefano Crump, Richard K.and Eusepi, Marc Giannoni, and Aysegul Sahini. A unified approach to measuring u<sup>\*</sup>. Technical report, NBER working paper 25930, 2019.
- Marco Del Negro, Mark Giannoni, and Frank Schorfheide. Inflation in the great recession and in new keynesian models. *American Economic Journals: Macroeconomics*, 7:168–196, 2015.
- Christopher Erceg, Luca Guerrieri, and Christopher Gust. Can long run restrictions identify technology shocks? *Journal of the European Economic Association*, 3:1237–1278, 2005.
- Francesco Furlanetto, Paolo Gelain, and Marzie Taheri-Sanjani. Output gap, monetary policy tradeoffs, and financial frictions. *Review of Economic Dynamics*, page forthcoming, 2020.
- Angelia Grant and Joshua Chan. A bayesian model comparison for trend-cycle decompositions of output. Journal of Money Credit and Banking, 49:525–552, 2017.
- James Hamilton. Why you should never use the hodrick and prescott filter. *Review of Economics* and *Statistics*, 100:831–843, 2018.
- Lars Hansen and Thomas Sargent. Formulating and estimating dynamic linear rational expectation models. *Journal of Economic Dynamics and Control*, 2:7–46, 1979.
- Alejandro Justiniano, Giorgio Primiceri, and Andrea Tambalotti. Is there a trade-off between output and inflation stabilization. *American Economic Journals: Macroeconomics*, 5:1–31, 2013.
- Gunes Kamber, James Morley, and Benjamin Wong. Intuitive and reliable estimates of the output gap from a beveridge and nelson filter. *Review of Economic and Statistics*, 100:550–566, 2018.
- Mariano Kulish and Adrian Pagan. Turning points and oscillatory cycles. Technical report, University of Sydney, manuscript, 2019.
- Thomas Laubach and John William. Measuring the natural rate of interest redux. Technical report, Brookings, working paper 15, 2015.
- Thomas Lubik, Christian Matthes, and Fabio Verona. Assessing us output fluctuations across time and frequencies. Technical report, Fed of Richmond working paper 19-6, 2019.
- Adrian Pagan. Business cycle issues: some reflection on a literature. Technical report, University of Sydney and CAMA., 2019.
- Valerie Ramey and Sarah Zubairy. Government spending multipliers in good times and in bad: Evidence from us historical data. *Journal of Political Economy*, 126:850–901, 2018.
- Stephanie Schmitt-Grohe and martin Uribe. Exchange rates and uncovered interest differentials: the role of permanent monetary shocks. Technical report, Colombia University, 2019.

Frank Smets and Rafael Wouters. Shocks and frictions in us business cycles: A bayesian dsge approach. American Economic Review, 97:585–606, 2007.

#### Appendix

#### Equations of the DGP

Flexible price economy

$$1 * a_t = calfa * rkf_t + (1 - calfa) * wf_t - 0 * (1 - calfa)a_t$$
(4)

$$zcapf_t = (1/(czcap/(1-czcap))) * rkf_t$$
(5)

$$rkf_t = wf_t + labf_t - kf_t \tag{6}$$

$$kf_t = kpf_{t-1} + zcapf_t \tag{7}$$

$$invef_{t} = \left(\frac{1}{1 + cbetabar * cgamma}\right) * (invef_{t-1} + cbetabar * cgamma * invef_{t+1} + \left(\frac{1}{cgamma^{2} * csadjcost}\right) * pkf_{t}\right) + qs_{t}$$

$$(8)$$

$$pkf_{t} = -rrf_{t} - 0 * b_{t} + \left(\frac{1}{(1 - chabb/cgamma)/(csigma * (1 + chabb/cgamma))}\right) * b_{t} + (crk/(crk + (1 - ctou))) * rkf_{t+1} + ((1 - ctou)/(crk + (1 - ctou))) * pkf_{t+1}$$
(9)

$$cf_t = (chabb/cgamma)/(1 + chabb/cgamma) * cf_{t-1} + (1/(1 + chabb/cgamma)) * cf_{t+1}$$

+ 
$$((csigma - 1) * cwhlc/(csigma * (1 + chabb/cgamma))) * (labf_t - labf_{t+1})$$

$$- (1 - chabb/cgamma)/(csigma * (1 + chabb/cgamma)) * (rrf_t + 0 * b_t) + b_t$$
(10)

$$yf_t = ccy * cf_t + ciy * invef_t + g_t + crkky * zcapf_t$$
(11)

$$yf_t = cfc * (calfa * kf_t + (1 - calfa) * labf_t + a_t)$$
(12)
$$(chabh/caamma)$$

$$wf_t = csigl * labf_t + (1/(1 - chabb/cgamma)) * cf_t - \frac{(chabb/cgamma)}{(1 - chabb/cgamma)} * cf_{t-1}$$
(13)

$$kpf_t = (1 - cikbar) * kpf_{t-1} + cikbar * invef_t + cikbar * (cgamma^2 * csadjcost) * qs_t$$
(14)

Sticky price - wage economy

$$mc_t = calfa * rk + (1 - calfa) * w_t - 1 * a_t - 0 * (1 - calfa) * a_t$$
(15)

$$zcap_t = (1/(czcap/(1-czcap))) * rk_t$$
(16)

$$rk_t = w_t + lab_t - k_t \tag{17}$$

$$k_t = k p_{t-1} + z c a p_t \tag{18}$$

 $inve_{t} = (1/(1 + cbetabar * cgamma)) * (inve_{t-1} + cbetabar * cgamma * inve_{t+1}$  $+ (1/(cgamma^{2} * csadjcost)) * pk_{t}) + qs_{t}$ (19)

$$pk_{t} = -r_{t} + pinf_{t+1} - 0 * b_{t} + (1/((1 - chabb/cgamma)/(csigma * (1 + chabb/cgamma)))) * b_{t} + (crk/(crk + (1 - ctou))) * rk_{t+1} + ((1 - ctou)/(crk + (1 - ctou))) * pk_{t+1}$$
(20)

Law of motion of shocks

$$a_t = crhoa * a_{t-1} + ea_t \tag{28}$$

$$b_t = crhob * b_{t-1} + eb_t \tag{29}$$

$$g_t = crhog * g_{t-1} + eg_t + cgy * ea_t \tag{30}$$

$$qs_t = crhoqs * qs_{t-1} + eqs_t \tag{31}$$

$$ms_t = crhoms * ms_{t-1} + em_t$$
(32)
$$min f_{t-1} = crhomin f_{t+1} min f_{t-1} + crim f_{t-1} ma_{t-1} ma_$$

$$spinf_{t} = crhopinf * spinf_{t-1} + epinfma_{t} - cmap * epinfma_{t-1}$$
(33)  
$$epinfma_{t} = epinf_{t}$$
(34)

$$epinfma_t = epinf_t$$

$$sw_t = crhow * sw_{t-1} + ewma_t - cmaw * ewma_{t-1}$$

$$(35)$$

$$ewma_t = ew_t$$

$$kp_t = (1 - cikbar) * kp_{t-1} + cikbar * inve_t + cikbar * cgamma^2 * csadjcost * qs_t$$
(37)

(36)

Tables and figures

Variable	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
					Ga	ар					
Y	27.36	25.57	29.07	48.5	27.1	28.41	27.6	23.14	10852	184.5	18.74(*)
С	29.46	32.42	35.35	45.74	33.85	33.53	32.79	32.44	NaN	NaN	19.63(*)
Ι	125.37	85.87	86.83	240.06	90.35	114.44	109.71	106.12	NaN	NaN	83.33(*)
Н	2.79	6.26	7.71	9.26	6.81	5.76	4.7	5.12	1.94	1.94	0.99(*)
RK	4.95	2.36	2.85	6.11	2.57	3.12	3.12	2.85	NaN	NaN	4.37
W	17.7	23.72	25.25	31.71	24.51	24.37	23.93	23.95	NaN	NaN	10.09(*)
CapU	15.67	7.48	9.03	19.37	8.15	9.89	9.88	9.07	NaN	NaN	13.86
$\pi$	0.17	0.45	0.65	0.65	0.53	0.45	0.26	0.52	NaN	NaN	0.03(*)
R	0.16	0.47	0.76	0.76	0.61	0.53	0.25	0.8	NaN	NaN	0.04(*)
Factor	2.48	4.61	5.07	4.64	4.83	4.02	4.15	5.02	NaN	NaN	1.83(*)
					Trans	sitory					
Y	74.71	36.46	33.96	75.35	35.81	47.45	49.98	37.1	24543	302.32	56.11
С	92.58	41.55	40.1	80.13	41.03	53.65	51.61	42.91	NaN	NaN	63
Ι	307.1	200.36	181.74	385.41	198.38	252.35	268.54	238.4	NaN	NaN	283.76
Н	16.01	10.37	9.67	18.38	10.12	12.3	13.62	9.78	521.27	521.27	14.28
RK	12.14	7.54	7.31	12.68	7.48	9.11	8.91	7.57	NaN	NaN	10.41
W	51.56	25.53	25.08	46.39	25.34	32.38	30.26	26.63	NaN	NaN	37.44
CapU	38.47	23.91	23.17	40.18	23.69	28.87	28.25	24.23	NaN	NaN	33
$\pi$	1.06	0.77	0.69	1.31	0.74	0.87	1	0.87	NaN	NaN	0.98
R	1.32	0.99	0.97	1.66	0.9	1.04	1.26	0.9	NaN	NaN	1.23
Factor	6.4	4.66	4.57	6.54	4.63	5.21	5.24	4.87	NaN	NaN	5.65

Table 4: Average MSE, T=750

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$ the inflation rate. Factor is the first principal component of the nine series. The MSE is computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
					Ga	ар					
Y	43.84	25.08	27.93	54.02	24.83	27.65	28.29	21.26	439.69	222.04	20.73(*)
C	53.13	33.03	35.1	52.69	33.4	36.33	35.48	34.36	NaN	NaN	18.44(*)
I	162.86	83.09	81.71	301.44	81.7	127.4	118.88	74.94	NaN	NaN	74.76(*)
H	5.35	6.55	7.71	9.1	6.51	5.36	4.91	6.32	2.06	2.06	1.04(*)
RK	6.7	3.35	3.75	6.1	3.27	3.11	3.84	3.19	NaN	NaN	3.95
W	39.03	26.31	28.01	38.73	26.33	27.07	27.63	21.12	NaN	NaN	10.72(*)
CapU	21.23	10.63	11.89	19.34	10.38	9.85	12.18	10.08	NaN	NaN	12.51
$\pi$	0.42	0.45	0.64	0.78	0.49	0.5	0.26	0.6	NaN	NaN	0.03(*)
R	0.4	0.49	0.76	0.81	0.61	0.51	0.25	0.93	NaN	NaN	0.04(*)
Factor	5.44	5.04	5.33	5.39	5.09	4.37	4.54	5.36	NaN	NaN	1.78(*)
					Trans	sitory					
Y	104.69	38.47	37.49	69.19	38.86	47.51	49.89	33.64	995.6	481.3	49.1
C	134.98	45.2	44.78	85.81	44.92	57.57	53.08	36.17	NaN	NaN	53.99
I	383.51	219.49	205.35	415.73	219.43	283.34	302.13	206.23	NaN	NaN	302.89
H	21.14	11.53	11.35	17.27	11.64	13.28	15.29	9.17	563.41	563.41	13.96
RK	17.42	9.57	9.42	14.1	9.58	10.96	11.29	9.03	NaN	NaN	10.49
W	78.62	21.8	21.52	45.11	21.89	29.38	25.85	26.11	NaN	NaN	27.64
capU	55.19	30.34	29.86	44.68	30.37	34.74	35.78	27.89	NaN	NaN	33.24
$\pi$	1.21	0.78	0.74	1.32	0.75	0.89	1.11	0.73	NaN	NaN	0.91
R	1.66	1.07	1.1	1.87	0.97	1.18	1.52	0.83	NaN	NaN	1.24
Factor	9.81	5.85	5.87	8.2	5.83	6.7	6.61	5.23	NaN	NaN	6.17

Table 5: Average real time MSE, T=750

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$ the inflation rate. Factor is the first principal component of the nine series. The MSE is computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
			Gap	s and	l Filt	ered	variał	oles			
Y	0.65	0.34	0.03	0.38	0.25	0.41	0.44	0.44	-0.33	0.25	0.71(*)
C	0.65	0.33	0.06	0.36	0.23	0.37	0.38	0.31	NaN	NaN	0.72(*)
I	0.59	0.27	0.04	0.34	0.19	0.37	0.44	0.43	NaN	NaN	0.65(*)
Н	0.86	0.42	0.06	0.5	0.31	0.54	0.68	0.67	0.93	0.93	0.93(*)
RK	0.53	0.41	0.04	0.38	0.29	0.4	0.36	0.22	NaN	NaN	0.55(*)
W	0.67	0.26	0.01	0.31	0.16	0.32	0.3	0.06	NaN	NaN	0.77(*)
CapU	0.53	0.41	0.04	0.38	0.29	0.4	0.36	0.16	NaN	NaN	0.55(*)
$\pi$	0.93	0.61	0.17	0.57	0.48	0.61	0.84	0.57	NaN	NaN	1.00(*)
R	0.94	0.67	0.29	0.59	0.48	0.6	0.88	0.36	NaN	NaN	1.00(*)
Factor	0.71	0.31	0.04	0.43	0.2	0.45	0.41	0.16	NaN	NaN	0.79(*)
		Т	ransit	ory a	and H	Filter	ed var	riable	s		
Y	0.01	0	0	0.01	0	0.01	-0.01	0.01	0.05	0	0.01
C	0.01	0	-0.01	-0.01	0	-0.02	-0.02	0.01	NaN	NaN	0
I	0.01	0	-0.01	0	0	0	-0.01	0	NaN	NaN	0
Н	-0.01	0	0	0.01	0	0.01	0	0	-0.02	-0.02	0
RK	0.01	0	-0.01	-0.01	0	-0.01	-0.01	0	NaN	NaN	0.01
W	-0.02	0.01	-0.01	-0.02	0	-0.02	-0.02	0	NaN	NaN	-0.01
CapU	0.01	0	-0.01	-0.01	0	-0.01	-0.01	-0.01	NaN	NaN	0.01
$\pi$	0.03	0	0	0.01	0	0.01	0.01	0.01	NaN	NaN	0.01
R	0.01	0	0	0	0	0	0	0	NaN	NaN	0
Factor	0.03	0	0	0	0	0	0	0.01	NaN	NaN	0.01

Table 6: Contemporaneous correlations, T=750

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. Correlations are computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	True	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
				(	Gaps	and	filter	ed va	riable	es		
Y	0.98	0.98	0.83	0.21	0.96	0.93	0.98	0.89	0.89	1	0.97	0.98(*)
C	0.98	0.99	0.83	0.23	0.97	0.93	0.99	0.89	0.88	NaN	NaN	0.98(*)
I	0.99	0.98	0.89	0.51	0.97	0.93	0.98	0.91	0.98	NaN	NaN	0.98(*)
Н	0.98	0.95	0.78	0.11	0.93	0.93	0.97	0.89	0.9	0.96	0.96	0.96
RK	0.97	0.99	0.83	0.21	0.97	0.94	0.99	0.89	0.93	NaN	NaN	0.99
W	0.98	0.99	0.82	0.12	0.98	0.94	0.99	0.87	0.94	NaN	NaN	0.99
CapU	0.97	0.99	0.83	0.21	0.97	0.94	0.99	0.89	0.91	NaN	NaN	0.99
π	0.94	0.93	0.78	0.16	0.9	0.92	0.96	0.88	0.83	NaN	NaN	0.94(*)
R	0.83	0.80	0.54	-0.21	0.76	0.91	0.95	0.77	0.47	NaN	NaN	0.85(*)
Factor	0.98	0.99	0.8	0.14	0.97	0.93	0.99	0.89	0.93	NaN	NaN	0.99(*)
				Tra	nsito	ry a	nd fil	tered	varia	bles		
Y	0.97	0.98	0.83	0.2	0.96	0.93	0.98	0.89	0.89	0.99	0.98	0.97(*)
C	0.99	0.99	0.84	0.26	0.98	0.93	0.99	0.89	0.87	NaN	NaN	0.98
I	0.98	0.97	0.89	0.5	0.97	0.93	0.98	0.91	0.98	NaN	NaN	0.97
Н	0.96	0.95	0.78	0.1	0.93	0.93	0.97	0.89	0.9	0.96	0.96	0.94
RK	0.99	0.98	0.83	0.19	0.97	0.94	0.99	0.89	0.93	NaN	NaN	0.98
W	0.99	0.99	0.82	0.14	0.98	0.94	0.99	0.87	0.94	NaN	NaN	0.98
CapU	0.99	0.98	0.83	0.19	0.97	0.94	0.99	0.89	0.93	NaN	NaN	0.98
$\pi$	0.94	0.93	0.78	0.15	0.91	0.92	0.97	0.88	0.84	NaN	NaN	0.91
R	0.84	0.81	0.54	-0.21	0.76	0.91	0.95	0.76	0.41	NaN	NaN	0.77
Factor	0.99	0.98	0.8	0.12	0.96	0.93	0.99	0.88	0.93	NaN	NaN	0.97

Table 7: AR1 coefficient, T=750

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. The AR1 coefficient is computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	True	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
					Ga	ps and	d filter	ed var	iables			
Y	23.62	38.81	4.07	1.56	42.88	3.45	15	16.62	6.83	4632.6	81.43	31.08
C	28.77	36.04	2.14	0.82	29.97	1.78	10.41	9.93	4.14	NaN	NaN	31.38
I	69.37	171.48	26.61	6.85	235.27	24.01	84.51	98.16	139.45	NaN	NaN	129.97
H	6.43	6.82	1.39	0.65	9.39	1.15	3.28	4.46	1.26	8.06	8.06	4.98
RK	2.48	6.3	0.4	0.15	6.05	0.33	2.1	1.88	0.3	NaN	NaN	5.21
W	20.85	20.83	0.9	0.38	17.78	0.68	6.13	4.67	0.6	NaN	NaN	18.04
CapU	7.86	19.98	1.25	0.48	19.17	1.03	6.64	5.96	0.8	NaN	NaN	16.52
$\pi$	0.58	0.47	0.14	0.07	0.68	0.12	0.24	0.39	0.25	NaN	NaN	0.35
R	0.69	0.59	0.24	0.22	0.88	0.15	0.27	0.51	0.17	NaN	NaN	0.42
Factor	5.05	3.13	0.21	0.09	2.8	0.17	0.97	0.93	0.39	NaN	NaN	2.7
					Trans	itory	and fil	tered v	variable	es		
Y	28.94	42.79	4.07	1.56	42.04	3.38	14.72	17.11	6.16	7649.1	140.59	24.2(*)
C	33.58	54.45	2.32	0.86	38.29	1.86	13.22	11.65	3.99	NaN	NaN	23.65
I	160.43	137.45	25.7	6.62	203.33	23.23	73.7	89.49	100.49	NaN	NaN	110.4
H	8.29	6.96	1.35	0.64	9.19	1.1	3.21	4.5	1.15	12.03	12.03	5.25
RK	6.32	5.14	0.38	0.15	5.33	0.31	1.85	1.72	0.3	NaN	NaN	3.37
W	20.43	26.62	0.93	0.39	19.92	0.7	6.85	5.06	0.74	NaN	NaN	12.45
CapU	20.02	16.3	1.2	0.46	16.88	0.97	5.86	5.45	1.01	NaN	NaN	10.69
$\pi$	0.57	0.46	0.14	0.06	0.67	0.11	0.23	0.37	0.25	NaN	NaN	0.37
R	0.69	0.58	0.23	0.21	0.87	0.15	0.27	0.5	0.15	NaN	NaN	0.49
Factor	4.52	2.11	0.17	0.08	1.98	0.14	0.69	0.72	0.44	NaN	NaN	1.18

Table 8: Variability, T=750

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. Y is output, C consumption, I investment, H hours, RR the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. The variability is computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	True	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
					Outpu	ıt Gap	and fil	tered o	output			
number TP	125.87	123.4	124.78	109.11	124.51	129.74	135.95	112.77	130.24	147.91	111.78	128
DurE	5.67	5.8	5.73	6.68	5.79	5.5	5.22	6.39	5.52	4.5	6.49	5.62(*)
DurR	5.71	5.81	5.75	6.49	5.75	5.5	5.27	6.32	5.49	5.19	6.36	5.57
AmpE	-2.49	-3.62	-3.57	-4.69	-5.09	-1.43	-1.47	-5.67	-1.48	-16.91	-6.47	-2.67(*)
AmpR	2.49	3.62	3.57	4.69	5.09	1.43	1.47	5.67	1.48	16.91	6.47	2.67(*)
			•		Facto	or Gap	and fil	tered f	actor			
Number TP	120.07	120.44	122.63	103.61	120.51	130.47	135.31	113.42	130.78	NaN	NaN	124.97
DurE	6.01	5.96	5.84	6.87	5.98	5.48	6.28	5.25	5.46	NaN	NaN	5.71
DurR	5.92	5.93	5.84	7.05	5.93	5.46	6.39	5.31	5.63	NaN	NaN	5.75
AmpE	-1.08	-0.83	-0.82	-1.12	-1.17	-0.3	-1.36	-0.32	-0.49	NaN	NaN	-0.61
AmpR	1.08	0.83	0.82	1.12	1.17	0.3	1.36	0.32	0.49	NaN	NaN	0.61
				Οι	נ tput	Fransito	ory and	filtere	ed outp	ut		
number TP	125.57	123.94	125.69	109.32	125.35	130.63	137.32	112.3	130.72	150.61	111.23	123.88
DurE	5.66	5.76	5.72	6.56	5.73	5.44	5.19	6.35	5.6	4.18	6.52	5.76
DurR	5.75	5.79	5.66	6.55	5.68	5.49	5.2	6.45	5.37	5.32	6.38	5.79(*)
AmpE	-3.4	-3.62	-3.58	-4.72	-5.11	-1.41	-1.47	-5.74	-1.44	-22.65	-5.59	-3.62(*)
AmpR	3.4	3.61	3.57	4.71	5.1	1.41	1.47	5.73	1.44	22.65	5.59	3
				F	'actor t	ransito	ory and	filtere	d facto	r		
Number TP	121.2	120.79	121.56	105.4	121.8	131.03	113.62	135.26	126.26	NaN	NaN	120.69
DurE	5.9	5.96	5.87	6.87	5.93	5.45	6.25	5.25	5.72	NaN	NaN	5.96
DurR	5.92	5.92	5.94	6.76	5.86	5.44	6.38	5.32	5.76	NaN	NaN	5.92(*)
AmpE	-0.98	-0.74	-0.73	-1	-1.05	-0.27	-1.17	-0.29	-0.52	NaN	NaN	-0.74
AmpR	0.98	0.74	0.73	1	1.05	0.27	1.17	0.29	0.52	NaN	NaN	0.74

Table 9: Number of turning points, average durations, average amplitudes, T=750

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW is the Butterworth filter. Y is output and Factor is the first principal component of the nine series. DurE and DurR are the durations of expansions and recessions; AmpE and AmpR the amplitude of expansions and recessions. Statistics are computed averaging over 100 data replications. In bold is the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Step ahead	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
		Phill	ips cu	rve p	redict	ion:	Outp	out G	ар		
1 step	0.25	0.26	0.27	0.37	0.26	0.32	0.36	0.27	0.36	0.4	0.21(*)
4 steps	1.41	1.5	1.67	1.91	1.67	1.55	1.87	1.53	2.14	2.23	1.04(*)
	Ph	illips	curve	e pred	iction	i: Tra	ansite	ory O	utput	t	
1 step	0.21	0.45	0.19	0.14	0.43	0.15	0.14	0.19	0.13	0.1	0.21
4 steps	0.79	0.93	0.6	0.49	2.89	0.7	0.52	0.77	0.52	0.58	0.68
		Ok	un la	w pre	dictic	n: O	utput	t Gap	)		
1 step	0.24	1.24	0.24	0.19	3.79	0.21	0.19	1.21	0.21	0.26	0.44
4 steps	3.67	33.84	2.49	1.75	45.94	2.74	2.23	10.88	2.43	2.04	3.13
	(	Okun	law p	oredic	tion:	tran	sitory	Out	put		
1 step	0.32	0.31	0.33	0.30	0.31	0.32	0.31	0.37	0.34	0.33	0.32
4 steps	3.02	2.81	2.76	2.68	3.05	2.73	2.85	3.27	3.21	2.89	2.93

Table 10: Phillips curve and Okun law predictions, T=750

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW is the Butterworth filter. The Phillips curve and the Okun law predictions are regression of the form:  $x_{t+m} = \alpha_0 + \alpha_1 x_t + \sum_{j=1}^3 \beta_j y_{t-j}$  where  $y_{t-j}$  is the true gap ( transitory) or the estimated one,  $x_t = \pi_t$  or  $h_t$  and m=1,4. Reported the difference in variance of the prediction error between each procedure and the true prediction error, averaged over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.



Figure 15: Estimated squared gain functions for output; unit root in TFP

#### On-line appendix

Variable	HPa	LDa	BPa	Hama	BNa	BQa	FODa	Trigo	BK	UCbiv
					Ga	ap				
Y	24.31	42.79	26.38	27.09	29.07	29.07	32.62	27.16	27.1	315.18
С	30.24	41.98	32.75	31.8	12.34	12.34	36.48	33.88	33.85	NaN
Ι	95.33	211.45	99.14	116.97	75.29	75.29	120.79	90.67	90.35	NaN
Н	5.29	9.11	6.14	3.98	NaN	NaN	8.43	6.82	6.81	NaN
RK	2.25	4.88	2.49	3.3	NaN	NaN	3.2	2.58	2.57	NaN
W	22.04	29.56	23.82	22.85	NaN	NaN	26.17	24.53	24.51	NaN
CapU	7.14	15.47	7.88	10.47	NaN	NaN	10.13	8.16	8.15	170.23
$\pi$	0.38	0.65	0.45	0.22	NaN	NaN	0.65	0.53	0.53	NaN
R	0.38	0.76	0.53	0.21	NaN	NaN	0.76	0.61	0.61	NaN
Factor	4.07	4.66	4.51	3.66	NaN	NaN	5.06	4.84	4.83	NaN
					Trans	sitory				
Y	41.68	65.77	39.94	56.77	33.96	33.96	41.32	35.83	35.81	119.31
С	45.84	67.11	44.14	58.69	34.14	34.14	44.75	41.05	41.03	NaN
Ι	230.26	357.78	222.85	293.58	198.98	198.98	229.67	198.61	198.38	NaN
Н	11.55	17.01	11.09	14.61	NaN	NaN	11.99	10.13	10.12	NaN
RK	8.19	11.13	7.94	9.73	NaN	NaN	8.04	7.48	7.48	NaN
W	27.7	38.84	26.78	33.99	NaN	NaN	27.06	25.34	25.34	NaN
CapU	25.94	35.26	25.17	30.83	NaN	NaN	25.46	23.7	23.69	113.85
$\pi$	0.85	1.26	0.81	1.04	NaN	NaN	0.93	0.74	0.74	NaN
R	1.08	1.62	0.98	1.3	NaN	NaN	1.27	0.9	0.9	NaN
Factor	4.89	6.02	4.81	5.55	NaN	NaN	4.88	4.63	4.63	NaN

Table A.1: Average MSE, T=750

Notes: HPa is the modified Hodrick and Prescott filtering, FODa is the forth order differencing, LDa is 16 quarter differencing, BPa is the modified band pass filtering, Hama is modified local projection detrending, BNa and BQa are trivariate Beveridge and Nelson and Blanchard and Quah decompositions, Trigo and BK are the trigonometric and the Baxter and King versions of a BP filter, UCbiv is a bivariate UC filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. Statistics are computed averaging over 100 data replications. For C and I BNa and BQa refer to the comparison of filtered C/Y and I/Y with the data. In bold are cases where the reported statistic improves the best result presented in table 4.

Variable	HPa	LDa	BPa	Hama	BNa	BQa	FODa	Trigo	BK	UCbiv
					G	ар				
Y	25.26	44.46	22.44	27.71	27.93	27.93	32.59	26.02	24.83	8.61
C	34.6	47.47	32.24	35.45	13.45	13.45	37.88	33.83	33.4	NaN
I	92.97	259.48	86.36	128.54	72.86	72.86	126.45	87.39	81.7	NaN
H	6.15	9.11	5.77	4.06	NaN	NaN	8.61	6.67	6.51	NaN
RK	3.3	5.17	3.12	3.71	NaN	NaN	4.09	3.32	3.27	NaN
W	27.01	34.22	24.85	25.74	NaN	NaN	30.06	27	26.33	NaN
capU	10.47	16.39	9.89	11.76	NaN	NaN	12.96	10.53	10.38	3.76
$\pi$	0.44	0.77	0.46	0.23	NaN	NaN	0.67	0.53	0.49	NaN
R	0.44	0.81	0.52	0.21	NaN	NaN	0.79	0.65	0.61	NaN
Factor	4.96	5.08	4.72	4	NaN	NaN	5.37	5.15	5.09	NaN
					Trans	sitory				
Y	40.83	62.42	39.27	55.83	37.49	37.49	42.85	38.65	38.86	7.52
C	47.14	68.28	<b>44.55</b>	58.09	49.59	49.59	48.62	45.08	44.92	NaN
I	236.45	397.07	237.89	345.75	230.58	230.58	243.9	218.85	219.46	NaN
H	11.92	17.11	11.58	16.05	NaN	NaN	13.17	11.63	11.65	NaN
RK	9.79	12.64	9.88	12.49	NaN	NaN	9.81	9.55	9.59	NaN
W	22.49	34.8	22.6	30.08	NaN	NaN	22.43	21.93	21.89	NaN
capU	31.02	40.04	31.29	39.58	NaN	NaN	31.1	30.25	30.37	2.06
$\pi$	0.84	1.3	0.79	1.13	NaN	NaN	1.03	0.78	0.75	NaN
R	1.13	1.7	1.01	1.52	NaN	NaN	1.42	1.01	0.97	NaN
Factor	6.04	7.18	5.85	6.78	NaN	NaN	6.18	5.8	5.83	NaN

Table A.2: Average real time MSE, T=750

Notes: HPa is the modified Hodrick and Prescott filtering, FODa is the forth order differencing, LDa is 16 quarter differencing, BPa is the modified band pass filtering, Hama is modified local projection detrending, BNa and BQa are trivariate Beveridge and Nelson and Blanchard and Quah decompositions, Trigo and BK are the trigonometric and the Baxter and King versions of a BP filter, UCbiv is a bivariate UC filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. Statistics are computed averaging over 100 data replications. For C and I BNa and BQa refer to the comparison of filtered C/Y and I/Y with the data. In bold are cases where the reported statistic improves the best result presented in table 5.

Variable	HPa	LDa	BPa	Hama	BNa	BQa	FODa	Trigo	BK	UCbiv
					(	Gap				
Y	0.46	0.35	0.36	0.53	0.03	0.03	0.15	0.24	0.25	-0.01
C	0.44	0.32	0.32	0.46	0.81	0.81	0.16	0.22	0.23	NaN
Ι	0.38	0.31	0.29	0.51	0.73	0.73	0.14	0.19	0.19	NaN
Н	0.58	0.45	0.46	0.76	NaN	NaN	0.22	0.31	0.31	NaN
RK	0.5	0.35	0.39	0.44	NaN	NaN	0.16	0.29	0.29	NaN
W	0.39	0.25	0.26	0.4	NaN	NaN	0.07	0.15	0.16	NaN
capU	0.5	0.35	0.39	0.44	NaN	NaN	0.16	0.29	0.29	0.03
$\pi$	0.71	0.55	0.6	0.88	NaN	NaN	0.38	0.47	0.48	NaN
R	0.76	0.57	0.59	0.9	NaN	NaN	0.44	0.48	0.48	NaN
Factor	0.46	0.38	0.32	0.52	NaN	NaN	0.15	0.2	0.2	NaN
					Tra	nsitor	'y			
Y	0	0	0	0	0	0	0	0	0	0.07
C	0	-0.02	0	-0.02	-0.05	-0.05	-0.01	0	0	NaN
I	0	0	0	-0.01	0.01	0.01	-0.01	0	0	NaN
Н	0	0.01	0	0	NaN	NaN	0	0	0	NaN
RK	0	-0.01	0	0	NaN	NaN	-0.01	0	0	NaN
W	0	-0.01	0.01	-0.03	NaN	NaN	-0.01	0	0	NaN
CapU	0	-0.01	0	0	NaN	NaN	-0.01	0	0	-0.01
$\pi$	0.01	0.01	0.01	0.01	NaN	NaN	0	0	0	NaN
R	0	-0.01	0	-0.01	NaN	NaN	0	0	0	NaN
Factor	0	0	0	0	NaN	NaN	0	0	0	NaN

Table A.3: Correlations, T=750

Notes: HPa is the modified Hodrick and Prescott filtering, FODa is the forth order differencing, LDa is 16 quarter differencing, BPa is the modified band pass filtering, Hama is modified local projection detrending, BNa and BQa are trivariate Beveridge and Nelson and Blanchard and Quah decompositions, Trigo and BK are the trigonometric and the Baxter and King versions of a BP filter, UCbiv is a bivariate UC filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. Statistics are computed averaging over 100 data replications. For C and I BNa and BQa refer to the comparison of filtered C/Y and I/Y with the data. In bold are cases where the reported statistic improves the best result presented in table 6.

Variable	True	HPa	LDa	BPa	Hama	BNa	BQa	FODa	Trigo	BK	UCbiv
						Ga	р				
Y	0.98	0.92	0.95	0.96	0.94	0.21	0.21	0.83	0.93	0.93	0.84
С	0.98	0.93	0.96	0.97	0.93	0.98	0.98	0.84	0.93	0.93	NaN
Ι	0.99	0.94	0.96	0.96	0.95	0.98	0.98	0.88	0.93	0.93	NaN
Η	0.98	0.88	0.92	0.95	0.92	NaN	NaN	0.78	0.93	0.93	NaN
RK	0.97	0.93	0.96	0.97	0.93	NaN	NaN	0.84	0.94	0.94	NaN
W	0.98	0.94	0.97	0.97	0.92	NaN	NaN	0.84	0.94	0.94	NaN
CapU	0.97	0.93	0.96	0.97	0.93	NaN	NaN	0.84	0.94	0.94	0.78
$\pi$	0.94	0.86	0.9	0.95	0.91	NaN	NaN	0.77	0.92	0.92	NaN
R	0.83	0.67	0.75	0.94	0.79	NaN	NaN	0.57	0.91	0.91	NaN
Factor	0.98	0.92	0.96	0.97	0.93	NaN	NaN	0.82	0.93	0.93	NaN
					Т	ransi	tory				
Y	0.97	0.92	0.95	0.96	0.93	0.2	0.2	0.82	0.93	0.93	0.96
С	0.99	0.94	0.97	0.97	0.93	0.98	0.98	0.85	0.93	0.93	NaN
Ι	0.98	0.94	0.96	0.96	0.95	0.98	0.98	0.87	0.93	0.93	NaN
Н	0.96	0.87	0.92	0.96	0.92	NaN	NaN	0.77	0.93	0.93	NaN
RK	0.99	0.93	0.96	0.97	0.93	NaN	NaN	0.83	0.94	0.94	NaN
W	0.99	0.94	0.97	0.97	0.92	NaN	NaN	0.84	0.94	0.94	NaN
CapU	0.99	0.93	0.96	0.97	0.93	NaN	NaN	0.83	0.94	0.94	0.94
$\pi$	0.94	0.86	0.9	0.95	0.91	NaN	NaN	0.77	0.92	0.92	NaN
R	0.84	0.67	0.74	0.94	0.79	NaN	NaN	0.57	0.91	0.91	NaN
Factor	0.99	0.91	0.95	0.96	0.93	NaN	NaN	0.8	0.93	0.93	NaN

Table A.4: AR1 coefficient, T=750

Notes: HPa is the modified Hodrick and Prescott filtering, FODa is the forth order differencing, LDa is 16 quarter differencing, BPa is the modified band pass filtering, Hama is modified local projection detrending, BNa and BQa are trivariate Beveridge and Nelson and Blanchard and Quah decompositions, Trigo and BK are the trigonometric and the Baxter and King versions of a BP filter, UCbiv is a bivariate UC filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. Statistics are computed averaging over 100 data replications. For C and I BNa and BQa refer to the comparison of filtered C/Y and I/Y with the data. In bold are cases where the reported statistic improves the best result presented in table 7.

Variable	True	HPa	LDa	BPa	Hama	BNa	BQa	FODa	Trigo	ΒK	UCbiv
						Gap					
Y	23.62	9.55	33.07	7.64	22.67	1.55	1.55	8.87	3.5	3.45	276.56
C	28.77	5.55	21.42	4.27	14.67	13.32	13.32	4.85	1.81	1.78	NaN
I	69.37	59.88	195.72	50.47	127.94	94.69	94.69	55.95	24.4	24.01	NaN
H	6.43	2.63	8.1	2.13	5.36	NaN	NaN	3.02	1.17	1.15	NaN
RK	2.48	1.12	4.28	0.81	2.84	NaN	NaN	0.9	0.33	0.33	NaN
W	20.85	2.92	12	2.09	7.6	NaN	NaN	2.11	0.7	0.68	NaN
CapU	7.86	3.54	13.57	2.57	8.99	NaN	NaN	2.85	1.05	1.03	127.04
$\pi$	0.58	0.23	0.65	0.2	0.43	NaN	NaN	0.31	0.12	0.12	NaN
R	0.69	0.33	0.86	0.23	0.54	NaN	NaN	0.52	0.15	0.15	NaN
Factor	5.05	0.53	2	0.41	1.35	NaN	NaN	0.47	0.17	0.17	NaN
					Tr	ansitor	'y				
Y	28.94	9.38	32.94	7.56	23.87	1.55	1.55	8.82	3.42	3.38	261.54
C	33.58	6.46	26.44	4.88	18.27	21.93	21.93	5.36	1.88	1.86	NaN
I	160.43	55.57	178.23	48	114.37	104.02	104.02	53.11	23.55	23.23	NaN
H	8.29	2.53	7.89	2.07	5.5	NaN	NaN	2.92	1.11	1.1	NaN
RK	6.32	1.03	3.87	0.78	2.54	NaN	NaN	0.85	0.31	0.31	NaN
W	20.43	3.1	13.2	2.24	8.49	NaN	NaN	2.22	0.71	0.7	NaN
CapU	20.02	3.26	12.25	2.49	8.04	NaN	NaN	2.68	0.99	0.97	89.91
$\pi$	0.57	0.22	0.63	0.19	0.41	NaN	NaN	0.3	0.11	0.11	NaN
R	0.69	0.32	0.84	0.23	0.53	NaN	NaN	0.51	0.15	0.15	NaN
Factor	4.52	0.41	1.5	0.32	1.03	NaN	NaN	0.38	0.14	0.14	NaN

Table A.5: Variability, T=750

Notes: HPa is the modified Hodrick and Prescott filtering, FODa is the forth order differencing, LDa is 16 quarter differencing, BPa is the modified band pass filtering, Hama is modified local projection detrending, BNa and BQa are trivariate Beveridge and Nelson and Blanchard and Quah decompositions, Trigo and BK are the trigonometric and the Baxter and King versions of a BP filter, UCbiv is a bivariate UC filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. Statistics are computed averaging over 100 data replications. For C and I BNa and BQa refer to the comparison of filtered C/Y and I/Y with the data. In bold are cases where the reported statistic improves the best result presented in table 8.

Variable	True	HPa	LDa	BPa	Hama	BNa	BQa	FODa	Trigo	BK	UCbiv
					Outpu	t Gap					
Number of TP	125.87	123.44	124.31	128.28	119.83	109.11	109.11	141.49	129.37	129.74	129.2
DurE	5.67	5.77	5.75	5.55	5.99	6.68	6.68	5.03	5.52	5.5	4.93
DurR	5.71	5.83	5.77	5.58	6	6.49	6.49	5.06	5.52	5.5	6.27
AmpE	-2.49	-3.61	-5.08	-1.45	-4.95	-4.69	-4.69	-5.22	-1.43	-1.43	-21.68
AmpR	2.49	3.61	5.08	1.45	4.95	4.69	4.69	5.22	1.43	1.43	- 21.69
					Fa	ctor G	lap				
Number of TP	120.07	120.69	121.06	128.56	118.84	NaN	NaN	138.45	129.98	130.44	NaN
DurE	6.01	5.93	5.9	5.57	6.05	NaN	NaN	5.16	5.49	5.49	NaN
DurR	5.92	5.93	5.96	5.53	6.05	NaN	NaN	5.16	5.49	5.46	NaN
AmpE	-1.08	-0.83	-1.16	-0.31	-1.25	NaN	NaN	-1.19	-0.3	-0.3	NaN
AmpR	1.08	0.83	1.16	0.31	1.25	NaN	NaN	1.19	0.3	0.3	NaN
					Trans	itory o	output				
Number of TP	125.57	123.94	124.84	128.87	119.01	109.32	109.32	142.5	130.56	130.63	127.76
DurE	5.66	5.79	5.8	5.53	6.09	6.56	6.56	5.03	5.46	5.44	5.5
DurR	5.75	5.76	5.68	5.55	5.97	6.55	6.55	5	5.47	5.49	5.81
AmpE	-3.4	-3.61	-5.1	-1.45	-5.17	-4.72	-4.72	-5.25	-1.41	-1.41	-12.25
AmpR	3.4	3.61	5.1	1.45	5.18	4.71	4.71	5.25	1.41	1.41	12.27
					Trans	sitory :	factor				
Number of TP	121.2	120.91	120.96	129.39	118.07	NaN	NaN	140.58	131.44	131.04	NaN
DurE	5.9	5.97	5.98	5.52	6.12	NaN	NaN	5.07	5.44	5.45	NaN
DurR	5.92	5.91	5.89	5.52	6.03	NaN	NaN	5.11	5.43	5.44	NaN
AmpE	-0.98	-0.74	-1.05	-0.27	-1.07	NaN	NaN	-1.06	-0.27	-0.27	NaN
AmpR	0.98	0.74	1.05	0.27	1.08	NaN	NaN	1.06	0.27	0.27	NaN

Table A.6: Number of turning points, average durations, average amplitudes, T=750

Notes: HPa is the modified Hodrick and Prescott filtering, FODa is the forth order differencing, LDa is 16 quarter differencing, BPa is the modified band pass filtering, Hama is modified local projection detrending, BNa and BQa are trivariate Beveridge and Nelson and Blanchard and Quah decompositions, Trigo and BK are the trigonometric and the Baxter and King versions of a BP filter, UCbiv is a bivariate UC filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. Statistics are computed averaging over 100 data replications. For C and I BNa and BQa refer to the comparison of filtered C/Y and I/Y with the data. In bold are cases where the reported statistic improves the best result presented in table 9.

Variable	HPa	LDa BPa	Hama	BNa	BQa	FODa	Trigo	BK	UCbiv
		Phillip	s curv	ve pre	dictio	on: Ou	itput	Gap	
1 step	0.21	$0.36\ 0.25$	0.34	0.27	0.27	0.32	0.26	0.26	0.32
4 steps	1.48	$1.81\ 1.88$	1.74	1.67	1.67	1.69	1.68	1.67	1.6
		Phillips c	urve p	oredic	tion:	Trans	itory	Outpi	ıt
1 step	0.36	$0.15\ 0.39$	0.13	0.19	0.19	0.16	0.42	0.43	0.15
4 steps	1.03	$0.54\ 3.21$	0.5	0.6	0.6	0.62	2.87	2.89	0.73
		Phillip	s curv	ve pre	dictio	on: Ou	itput	Gap	
1 step	0.55	$0.21\ 4.67$	0.21	0.24	0.24	0.23	3.75	3.79	4.91
4 steps	17.25	$1.99\ 55.2$	2.56	2.49	2.49	2.27	45.59	45.94	17.28
		Okun la	w pre	edictio	on: T	ransit	ory Ou	ıtput	
1 step	0.32	$0.31\ 0.32$	0.31	0.33	0.33	0.31	0.31	0.31	0.34
4 steps	2.85	$2.69\ 3.21$	2.97	2.76	2.76	2.63	3.03	3.05	2.95

Table A.7 Phillips curve and Okun law predictions, T=750

Notes: HPa is the modified Hodrick and Prescott filtering, FODa is the forth order differencing, LDa is 16 quarter differencing, BPa is the modified band pass filtering, Hama is modified local projection detrending, BNa and BQa are trivariate Beveridge and Nelson and Blanchard and Quah decompositions, Trigo and Bk are the trigonometric and the Baxter and King versions of a BP filter, UCbiv is a bivariate UC filter. The Phillips curve and the Okun law predictions are regression of the form ;  $x_{t+m} = \alpha_0 + \alpha_1 x_t + \sum_{j=1}^3 \beta_j y_{t-j}$  where  $y_{t-j}$  is the true gap (transitory) or the estimated one,  $x_t = \pi_t$  or  $h_t$  and m=1,4. Reported is the difference in variance of the prediction error between each procedure and the true prediction error, averaged over 100 data replications. In bold cases where the reported statistic improves the best result presented in table 10.

Variable	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
					G	lap					
Y	27.26	26.32	29.77	51.56	27.55	29.84	28.02	104.64	1146.1	140.71	20.06(*)
C	31.97	33.63	36.48	50.27	34.8	36.07	34.25	112.85	NaN	NaN	22.36(*)
I	111.95	86	87.82	250.93	89.22	117.93	107.82	159.06	NaN	NaN	90
H	5.51	6.58	7.93	8.9	7.02	5.69	5.27	89.77	3.58	3.58	1.08(*)
RK	2.63	<b>2.5</b>	3.02	6.64	2.69	3.54	3.19	2.71	NaN	NaN	4.76
W	19.78	21.09	22.65	30.86	21.79	22.57	21.86	18.93	NaN	NaN	11.04(*)
CapU	8.32	7.92	9.58	21.03	8.51	11.22	10.09	22.82	NaN	NaN	15.08
$\pi$	0.42	0.51	0.69	0.66	0.57	0.48	0.37	0.59	NaN	NaN	0.03(*)
R	0.4	0.5	0.79	0.76	0.63	0.54	0.35	0.6	NaN	NaN	0.04(*)
Factor	4.01	4.5	5.21	6.56	4.77	4.36	4.4	4.72	NaN	NaN	2.49(*)
					Tran	sitory					
Y	47.3	37.73	35.29	77.94	36.87	49.42	48.31	87.87	1849.5	120.7	50.38
C	52.96	44.44	<b>43.08</b>	80.37	43.87	55.58	52.1	101.61	NaN	NaN	55.32
I	263.42	210.79	192.59	403.39	207.75	267.84	263.75	172.93	NaN	NaN	280.51
H	12.5	10.31	9.61	19.49	10.04	12.84	13.02	62.44	53.43	53.43	13.38
RK	9.09	7.52	7.31	12.45	7.44	9.07	8.54	3.16	NaN	NaN	9.59
W	33.57	28.47	28.05	48.54	28.21	35.16	31.96	23.57	NaN	NaN	36.51
CapU	28.82	23.83	23.16	39.46	23.58	28.74	27.06	21.37	NaN	NaN	30.38
$\pi$	0.88	0.77	0.71	1.3	0.75	0.86	0.91	0.79	NaN	NaN	0.93
R	1.12	0.99	0.99	1.68	0.91	1.06	1.19	0.86	NaN	NaN	1.2
Factor	6.25	4.74	4.39	10.55	4.61	6.53	6.14	7.91	NaN	NaN	6.89

Table A.8: Average MSE, T=150

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$ the inflation rate. Factor is the first principal component of the nine series. The MSE is computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
					Ga	p					
Y	36.98	35.83	38.66	42.87	36.09	26.59	33.48	120.31	334.35	269.97	18.85(*)
C	45.82	44.24	46.91	40.11	44.31	34.39	44.88	129.96	NaN	NaN	21.2(*)
Ι	134.64	110.61	111.67	248.47	110.73	115.56	117.75	287.04	NaN	NaN	78.04(*)
Н	9.46	8.98	10.28	9.09	9.05	6.39	5.98	82.42	3.41	3.41	1.04(*)
RK	3.83	2.41	2.69	8.21	2.44	3.89	3.5	7.61	NaN	NaN	5.24
W	24.24	22.62	23.18	27.01	22.66	20.14	23.12	29.28	NaN	NaN	10.19(*)
CapU	12.15	7.62	8.52	26	7.72	12.34	11.08	37.28	NaN	NaN	16.62
$\pi$	0.68	0.65	0.87	0.8	0.67	0.6	0.49	0.67	NaN	NaN	0.04(*)
R	0.68	0.65	0.93	0.88	0.74	0.64	0.45	0.78	NaN	NaN	0.05(*)
Factor	10.19	8.01	8.21	9.51	7.95	7.43	8.22	9.33	NaN	NaN	3.81(*)
					Transi	itory					
Y	45.68	39.3	38.8	76.63	39.31	49.81	49.64	122.66	459.47	217.52	52.81
C	53.27	44.02	43.52	89.39	44.09	59.56	52.25	127.17	NaN	NaN	56.19
Ι	231.54	187.57	173.41	377.49	187.57	250.4	241.36	279.46	NaN	NaN	244.74
Н	11.27	10.02	9.98	16.49	9.83	11.2	12.54	61.70	61.36	61.36	11.8
RK	9.74	7.7	7.64	14.38	7.56	10.04	9.1	5.41	NaN	NaN	8.05
W	38.01	32.35	32.08	58.36	32.32	42.02	35.64	37.61	NaN	NaN	40.23
CapU	30.85	24.41	24.22	45.56	23.96	31.81	28.82	25.24	NaN	NaN	25.52
$\pi$	0.83	0.73	0.69	1.26	0.73	0.86	0.95	0.75	NaN	NaN	0.91
R	1.11	0.92	0.92	1.8	0.83	1.11	1.34	0.75	NaN	NaN	1.3
Factor	8.41	6.44	6.38	11.44	6.42	8.05	8.02	7.22	NaN	NaN	7.24

Table A.9: Average real time MSE

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$ the inflation rate. Factor is the first principal component of the nine series. The MSE is computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
			Gap	s and	Filte	ered v	variab	les			
Y	0.59	0.45	0.04	0.48	0.33	0.52	0.54	0.13	-0.58	0.15	0.74(*)
С	0.58	0.45	0.08	0.46	0.33	0.48	0.48	0.11	NaN	NaN	0.76(*)
Ι	0.51	0.37	0.04	0.45	0.27	0.49	0.51	0.29	NaN	NaN	0.66(*)
Н	0.75	0.53	0.09	0.62	0.41	0.68	0.77	0.1	0.93	0.93	0.93(*)
RK	0.6	0.5	0.03	0.42	0.37	0.45	0.43	0.34	NaN	NaN	0.61(*)
W	0.57	0.4	0.02	0.44	0.26	0.45	0.39	0.13	NaN	NaN	0.74(*)
CapU	0.6	0.5	0.03	0.42	0.37	0.45	0.43	0.25	NaN	NaN	0.61(*)
$\pi$	0.87	0.74	0.22	0.65	0.6	0.72	0.91	0.58	NaN	NaN	1(*)
R	0.89	0.79	0.34	0.66	0.59	0.68	0.93	0.77	NaN	NaN	1(*)
Factor	0.49	0.34	0.01	0.38	0.24	0.41	0.39	0.32	NaN	NaN	0.7(*)
		Г	ransit	ory a	nd F	iltere	d vari	iables	5		
Y	-0.01	-0.01	-0.02	-0.02	-0.01	-0.03	-0.03	-0.02	-0.08	0.02	-0.03
С	0	0.01	-0.01	-0.04	0.02	-0.03	-0.02	-0.03	NaN	NaN	0
Ι	-0.03	0	-0.02	-0.02	-0.01	-0.01	-0.02	-0.03	NaN	NaN	0
Η	-0.02	-0.01	-0.02	-0.02	-0.01	-0.02	-0.03	0	-0.02	-0.02	-0.02
RK	-0.04	-0.01	-0.01	-0.02	-0.01	-0.01	-0.01	0.02	NaN	NaN	-0.04
W	-0.02	0.01	-0.03	-0.06	0.01	-0.05	-0.04	0.02	NaN	NaN	-0.06
CapU	-0.04	-0.01	-0.01	-0.02	-0.01	-0.01	-0.01	0	NaN	NaN	-0.04
$\pi$	0.01	0.01	0	0	0.01	0.02	0.03	-0.01	NaN	NaN	-0.01
R	0.01	0.01	0	-0.01	0.01	-0.01	0.02	0.01	NaN	NaN	0
Factor	-0.04	-0.01	-0.01	-0.06	-0.01	-0.06	-0.04	-0.01	NaN	NaN	-0.04

Table A.10: Contemporaneous correlations, T=150

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. Correlations are computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	True	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
			C	laps a	and fil	tered	vari	ables				
Y	0.96	0.94	0.81	0.2	0.95	0.93	0.98	0.89	1.00	0.97	0.79	0.96(*)
С	0.96	0.95	0.81	0.21	0.96	0.93	0.98	0.88	1.00	NaN	NaN	0.97
Ι	0.98	0.96	0.88	0.5	0.97	0.93	0.98	0.91	0.99	NaN	NaN	0.97
Н	0.96	0.9	0.76	0.11	0.92	0.92	0.97	0.88	1.00	0.93	0.93	0.93
RK	0.95	0.95	0.81	0.19	0.96	0.93	0.98	0.88	0.92	NaN	NaN	0.98
W	0.95	0.96	0.8	0.1	0.97	0.93	0.99	0.86	0.93	NaN	NaN	0.98
CapU	0.95	0.95	0.81	0.19	0.96	0.93	0.98	0.88	0.92	NaN	NaN	0.98
$\pi$	0.9	0.87	0.76	0.16	0.88	0.92	0.96	0.86	0.77	NaN	NaN	0.9 (*)
R	0.76	0.69	0.53	-0.21	0.73	0.91	0.94	0.72	0.29	NaN	NaN	0.78(*)
Factor	0.97	0.95	0.81	0.19	0.96	0.93	0.98	0.88	0.91	NaN	NaN	0.98(*)
			Tra	nsitor	y and	filte	red v	ariab	les			
Y	0.95	0.93	0.81	0.2	0.95	0.93	0.98	0.89	1.00	0.95	0.86	0.94
С	0.96	0.95	0.83	0.24	0.97	0.93	0.98	0.88	1.00	NaN	NaN	0.96(*)
Ι	0.96	0.95	0.88	0.48	0.96	0.93	0.98	0.9	0.99	NaN	NaN	0.96(*)
Н	0.94	0.89	0.76	0.09	0.92	0.92	0.97	0.87	0.99	0.93	0.93	0.91
RK	0.97	0.95	0.81	0.18	0.96	0.93	0.98	0.88	0.93	NaN	NaN	0.95
W	0.97	0.95	0.81	0.11	0.97	0.93	0.99	0.87	0.92	NaN	NaN	0.96
CapU	0.97	0.95	0.81	0.18	0.96	0.93	0.98	0.88	0.92	NaN	NaN	0.95
$\pi$	0.9	0.86	0.76	0.13	0.89	0.92	0.96	0.86	0.77	NaN	NaN	0.88
R	0.75	0.69	0.52	-0.21	0.73	0.91	0.94	0.72	0.29	NaN	NaN	0.73(*)
Factor	0.97	0.93	0.81	0.17	0.96	0.93	0.98	0.88	0.91	NaN	NaN	0.95

Table A.11: AR1 coefficient, T=150

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$ the inflation rate. Factor is the first principal component of the nine series. The AR1 coefficient is computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	True	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
				Gaps	s and f	iltered	l varia	ables				
Y	15.65	14.85	3.95	1.56	36.28	3.21	12.1	13.7	192.03	425.21	24.07	20.88
С	17.59	9.23	2.06	0.81	24.54	1.67	8.11	7.98	190.93	NaN	NaN	18.35(*)
Ι	44.67	87.39	25.58	6.85	205.24	22.25	70.02	80.12	242.9	NaN	NaN	93.09
Η	4.66	3.4	1.34	0.65	7.72	1.08	2.58	3.58	186.35	5.41	5.41	3.54
RK	1.9	1.92	0.38	0.15	4.87	0.3	1.61	1.53	0.23	NaN	NaN	2.65
W	11.4	5.54	0.86	0.38	14.81	0.64	4.81	3.75	0.51	NaN	NaN	9.36
CapU	6.02	6.07	1.2	0.48	15.44	0.94	5.12	4.85	29.82	NaN	NaN	8.41
$\pi$	0.42	0.27	0.14	0.07	0.54	0.11	0.18	0.3	0.08	NaN	NaN	0.26
R	0.53	0.38	0.23	0.22	0.75	0.15	0.22	0.41	0.06	NaN	NaN	0.32
Factor	5.36	1.92	0.49	0.27	4.82	0.4	1.56	1.7	0.26	NaN	NaN	2.8
			Ti	ransit	ory and	d filte	red va	ariable	es			
Y	18.21	13.18	3.93	1.57	33.83	3.1	11.28	13.14	204.37	652.08	25.43	19.44(*)
С	17.43	10.23	2.28	0.87	27.95	1.75	9.1	8.86	207.63	NaN	NaN	15.46(*)
Ι	113.84	70.9	24.77	6.64	168.26	21.36	58.08	68.56	237.58	NaN	NaN	102.97(*)
Н	5.68	3.39	1.31	0.65	7.98	1.04	2.61	3.55	169.43	6.93	6.93	4.72(*)
RK	3.95	1.62	0.36	0.14	4.02	0.29	1.32	1.31	0.23	NaN	NaN	2.2
W	10.97	5.44	0.89	0.38	15.03	0.66	4.85	3.83	0.45	NaN	NaN	8.79(*)
CapU	12.52	5.15	1.14	0.46	12.75	0.91	4.18	4.14	22.5	NaN	NaN	6.99
pi	0.38	0.25	0.13	0.06	0.53	0.1	0.17	0.27	0.08	NaN	NaN	0.32(*)
R	0.5	0.36	0.22	0.21	0.71	0.14	0.21	0.4	0.06	NaN	NaN	0.45(*)
Factor	4.53	1.82	0.54	0.27	4.58	0.43	1.51	1.72	4.38	NaN	NaN	2.64

Table A.12: Variability, T=150

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW the Butterworth filter. Y is output, C consumption, I investment, H hours, RK the real rate, W the real wage, CapU capacity utilization, R the nominal rate, and  $\pi$  the inflation rate. Factor is the first principal component of the nine series. The variability is computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Variable	True	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
			(	Dutput	Gap a	and fil	tered	outpu	t			
Number of TI	P 19.97	20.44	20.56	17.9	19.66	20.36	21.63	19.47	20.07	22.12	17.18	21.08
DurE	5.94	5.75	5.75	6.91	6.19	5.64	5.34	6	5.64	5.33	6.81	5.61
DurR	5.75	5.61	5.57	6.35	5.66	5.51	5.24	5.83	5.64	5.31	6.9	5.38
AmpE	-2.47	-3.58	-3.55	-4.7	-5.12	-1.41	-1.48	-5.29	-1.83	-11.88	-5.9	-2.66(*)
AmpR	2.46	3.59	3.55	4.72	5.1	1.4	1.47	5.27	1.81	11.86	5.92	2.67(*)
				factor	Gap a	nd filt	ered f	actor				
Number of TI	P 19.37	19.08	19.23	17.73	19.71	20.23	18.5	21.48	21.68	NaN	NaN	19.89
DurE	6.06	6.01	6.02	6.77	6.15	5.66	6.32	5.37	5.24			5.8
DurR	5.92	6.13	5.94	6.55	5.79	5.54	6.26	5.25	5.69	NaN	NaN	5.84
AmpE	-1.53	-1.21	-1.19	-1.55	-1.68	-0.46	-1.82	-0.48	-0.46	NaN	NaN	-0.89
AmpR	1.53	1.21	1.19	1.55	1.68	0.46	1.83	0.48	0.46	NaN	NaN	0.89
			Out	put Tr	ansitor	y and	filter	ed out	put			
Number of TI	21.05	19.99	20.11	17.67	19.6	20.47	22.04	19.05	20.44	22.77	17.87	19.85
DurE	5.47	5.66	5.63	6.85	5.98	5.48	5.08	6.34	5.82	4.7	6.73	5.72
DurR	5.57	6.01	5.84	6.38	5.85	5.61	5.18	6.16	5.61	5.63	6.8	6.06
AmpE	-3.42	-3.6	-3.56	-4.75	-5.18	-1.42	-1.47	-5.25	-1.84	-15.51	-5.46	-3.61
AmpR	3.42	3.61	3.57	4.74	5.17	1.42	1.46	5.24	1.81	15.46	5.48	3.61
			Fac	ctor tra	ansitor	y and	filtere	ed fact	tor			
Number of TI	P 19.97	19.07	19.1	17.85	19.59	20.33	19.02	21.68	21.84	NaN	NaN	18.98
DurE	5.91	6.14	6.14	6.62	6.04	5.58	6.05	5.28	5.24	NaN	NaN	6.2
DurR	5.73	6.02	5.97	6.44	5.71	5.61	6.26	5.26	5.39	NaN	NaN	6.03
AmpE	-1.34	-1.27	-1.26	-1.66	-1.8	-0.49	-1.87	-0.5	-1.41	NaN	NaN	-1.28(*)
AmpR	1.34	1.27	1.26	1.67	1.81	0.5	1.88	0.5	1.41	NaN	NaN	1.28(*)

Table A.13: Number of turning points, average durations, average amplitudes, T=150

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW is the Butterworth filter. Y is output and Factor is the first principal component of the nine series. DurE and DurR are the durations of expansions and recessions; AmpE and AmpR the amplitude of expansions and recessions. Statistics are computed averaging over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.

Step ahead	POLY	HP	FOD	LD	BP	Wa	Ham	UC	BN	BQ	BW
	Phillips curve prediction: Output Gap										
1 step	0.46	0.6	0.43	0.48	0.64	0.44	0.49	0.36	0.42	0.52	0.35(*)
4 steps	2.49	2.76	2.06	2.75	3.92	2.5	2.55	2.08	2.8	2.93	1.65(*)
		Phillips curve prediction: Transitory Output									
1step	0.58	0.64	0.53	0.45	0.53	0.5	0.55	0.61	0.49	0.52	0.57
4step	2.52	2.33	2.22	2.16	3.26	2.39	2.24	2.6	2.28	2.21	2.45
	Okun law prediction: Output Gap										
1step	0.79	1.21	0.75	0.71	3.26	0.68	0.69	0.7	0.84	0.69	0.84
4step	12.92	26.59	7.96	8.98	39.17	8.92	8.25	8.31	14.3	11.56	9.36
	Okun law prediction: transitory Output										
1step	1.46	1.44	1.53	1.39	1.45	1.38	1.53	1.45	1.51	1.45	1.48
4step	13.94	13.69	12.89	13.54	15.62	15.03	15.63	16.69	15.78	13.5	15.48

Table A.14: Phillips curve and Okun law predictions, T=150

Notes: POLY is polynomial detrending, HP is Hodrick and Prescott filtering, FOD is first order differencing, LD is 24 quarter differencing, UC is an unobservable component model, BP is band pass filtering, Ham is local projection detrending, BN and BQ are bivariate Beveridge and Nelson and Blanchard and Quah decompositions, BW is the Butterworth filter. The Phillips curve and the Okun law predictions are regression of the form ;  $x_{t+m} = \alpha_0 + \alpha_1 x_t + \sum_{j=1}^3 \beta_j y_{t-j}$  where  $y_{t-j}$  is the true gap ( transitory) or the estimated one,  $x_t = \pi_t$  or  $h_t$  and m=1,4. Reported the difference in variance of the prediction error between each procedure and the true prediction error, averaged over 100 data replications. In bold the best approach. A (\*) in the last column indicates that the BW filter improves over or is comparable to the best approach.



Figure A.1: Log spectra of gaps and of Butterworth filtered data



Figure A.2: Time series of gaps and of Butterworth filtered data



Figure A.3: Log spectra of transitory components and of Butterworth filtered data



Figure A.4: Time series of transitory data and of Butterworth filtered data

## Earlier Working Papers:

For a complete list of Working Papers published by Sveriges Riksbank, see www.riksbank.se

Estimation of an Adaptive Stock Market Model with Heterogeneous Agents by Henrik Amilon	2005:177
Some Further Evidence on Interest-Rate Smoothing: The Role of Measurement Errors in the Output Gap by Mikael Apel and Per Jansson	2005:178
Bayesian Estimation of an Open Economy DSGE Model with Incomplete Pass-Through by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:179
Are Constant Interest Rate Forecasts Modest Interventions? Evidence from an Estimated Open Economy DSGE Model of the Euro Area by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:180
Inference in Vector Autoregressive Models with an Informative Prior on the Steady State by Mattias Villani	2005:181
Bank Mergers, Competition and Liquidity by Elena Carletti, Philipp Hartmann and Giancarlo Spagnolo	2005:182
Testing Near-Rationality using Detailed Survey Data by Michael F. Bryan and Stefan Palmqvist	2005:183
Exploring Interactions between Real Activity and the Financial Stance by Tor Jacobson, Jesper Lindé and Kasper Roszbach	2005:184
Two-Sided Network Effects, Bank Interchange Fees, and the Allocation of Fixed Costs by Mats A. Bergman	2005:185
Trade Deficits in the Baltic States: How Long Will the Party Last? by Rudolfs Bems and Kristian Jönsson	2005:186
Real Exchange Rate and Consumption Fluctuations follwing Trade Liberalization by Kristian Jönsson	2005:187
Modern Forecasting Models in Action: Improving Macroeconomic Analyses at Central Banks by Malin Adolfson, Michael K. Andersson, Jesper Lindé, Mattias Villani and Anders Vredin	2005:188
Bayesian Inference of General Linear Restrictions on the Cointegration Space by Mattias Villani	2005:189
Forecasting Performance of an Open Economy Dynamic Stochastic General Equilibrium Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:190
Forecast Combination and Model Averaging using Predictive Measures by Jana Eklund and Sune Karlsson	2005:191
Swedish Intervention and the Krona Float, 1993-2002 by Owen F. Humpage and Javiera Ragnartz	2006:192
A Simultaneous Model of the Swedish Krona, the US Dollar and the Euro by Hans Lindblad and Peter Sellin	2006:193
Testing Theories of Job Creation: Does Supply Create Its Own Demand? by Mikael Carlsson, Stefan Eriksson and Nils Gottfries	2006:194
Down or Out: Assessing The Welfare Costs of Household Investment Mistakes by Laurent E. Calvet, John Y. Campbell and Paolo Sodini	2006:195
Efficient Bayesian Inference for Multiple Change-Point and Mixture Innovation Models by Paolo Giordani and Robert Kohn	2006:196
Derivation and Estimation of a New Keynesian Phillips Curve in a Small Open Economy by Karolina Holmberg	2006:197
Technology Shocks and the Labour-Input Response: Evidence from Firm-Level Data by Mikael Carlsson and Jon Smedsaas	2006:198
Monetary Policy and Staggered Wage Bargaining when Prices are Sticky by Mikael Carlsson and Andreas Westermark	2006:199
The Swedish External Position and the Krona by Philip R. Lane	2006:200

Price Setting Transactions and the Role of Denominating Currency in FX Markets by Richard Friberg and Fredrik Wilander	2007:201
The geography of asset holdings: Evidence from Sweden by Nicolas Coeurdacier and Philippe Martin	2007:202
Evaluating An Estimated New Keynesian Small Open Economy Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2007:203
The Use of Cash and the Size of the Shadow Economy in Sweden by Gabriela Guibourg and Björn Segendorf	2007:204
Bank supervision Russian style: Evidence of conflicts between micro- and macro-prudential concerns by Sophie Claeys and Koen Schoors	2007:205
Optimal Monetary Policy under Downward Nominal Wage Rigidity by Mikael Carlsson and Andreas Westermark	2007:206
Financial Structure, Managerial Compensation and Monitoring by Vittoria Cerasi and Sonja Daltung	2007:207
Financial Frictions, Investment and Tobin's q by Guido Lorenzoni and Karl Walentin	2007:208
Sticky Information vs Sticky Prices: A Horse Race in a DSGE Framework by Mathias Trabandt	2007:209
Acquisition versus greenfield: The impact of the mode of foreign bank entry on information and bank lending rates by Sophie Claeys and Christa Hainz	2007:210
Nonparametric Regression Density Estimation Using Smoothly Varying Normal Mixtures by Mattias Villani, Robert Kohn and Paolo Giordani	2007:211
The Costs of Paying – Private and Social Costs of Cash and Card by Mats Bergman, Gabriella Guibourg and Björn Segendorf	2007:212
Using a New Open Economy Macroeconomics model to make real nominal exchange rate forecasts by Peter Sellin	2007:213
Introducing Financial Frictions and Unemployment into a Small Open Economy Model by Lawrence J. Christiano, Mathias Trabandt and Karl Walentin	2007:214
Earnings Inequality and the Equity Premium by Karl Walentin	2007:215
Bayesian forecast combination for VAR models by Michael K. Andersson and Sune Karlsson	2007:216
Do Central Banks React to House Prices? by Daria Finocchiaro and Virginia Queijo von Heideken	2007:217
The Riksbank's Forecasting Performance by Michael K. Andersson, Gustav Karlsson and Josef Svensson	2007:218
Macroeconomic Impact on Expected Default Freqency by Per Åsberg and Hovick Shahnazarian	2008:219
Monetary Policy Regimes and the Volatility of Long-Term Interest Rates by Virginia Queijo von Heideken	2008:220
Governing the Governors: A Clinical Study of Central Banks by Lars Frisell, Kasper Roszbach and Giancarlo Spagnolo	2008:221
The Monetary Policy Decision-Making Process and the Term Structure of Interest Rates by Hans Dillén	2008:222
How Important are Financial Frictions in the US and the Euro Area by Virginia Queijo von Heideken	2008:223
Block Kalman filtering for large-scale DSGE models by Ingvar Strid and Karl Walentin	2008:224
Optimal Monetary Policy in an Operational Medium-Sized DSGE Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Lars E. O. Svensson	2008:225
Firm Default and Aggregate Fluctuations by Tor Jacobson, Rikard Kindell, Jesper Lindé and Kasper Roszbach	2008:226
Re-Evaluating Swedish Membership in EMU: Evidence from an Estimated Model by Ulf Söderström	2008:227

The Effect of Cash Flow on Investment: An Empirical Test of the Balance Sheet Channel by Ola Melander	2009:228
Expectation Driven Business Cycles with Limited Enforcement by Karl Walentin	2009:229
Effects of Organizational Change on Firm Productivity by Christina Håkanson	2009:230
Evaluating Microfoundations for Aggregate Price Rigidities: Evidence from Matched Firm-Level Data on Product Prices and Unit Labor Cost by Mikael Carlsson and Oskar Nordström Skans	2009:231
Monetary Policy Trade-Offs in an Estimated Open-Economy DSGE Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Lars E. O. Svensson	2009:232
Flexible Modeling of Conditional Distributions Using Smooth Mixtures of Asymmetric Student T Densities by Feng Li, Mattias Villani and Robert Kohn	2009:233
Forecasting Macroeconomic Time Series with Locally Adaptive Signal Extraction by Paolo Giordani and Mattias Villani	2009:234
Evaluating Monetary Policy by Lars E. O. Svensson	2009:235
Risk Premiums and Macroeconomic Dynamics in a Heterogeneous Agent Model by Ferre De Graeve, Maarten Dossche, Marina Emiris, Henri Sneessens and Raf Wouters	2010:236
Picking the Brains of MPC Members by Mikael Apel, Carl Andreas Claussen and Petra Lennartsdotter	2010:237
Involuntary Unemployment and the Business Cycle by Lawrence J. Christiano, Mathias Trabandt and Karl Walentin	2010:238
Housing collateral and the monetary transmission mechanism by Karl Walentin and Peter Sellin	2010:239
The Discursive Dilemma in Monetary Policy by Carl Andreas Claussen and Øistein Røisland	2010:240
Monetary Regime Change and Business Cycles by Vasco Cúrdia and Daria Finocchiaro	2010:241
Bayesian Inference in Structural Second-Price common Value Auctions by Bertil Wegmann and Mattias Villani	2010:242
Equilibrium asset prices and the wealth distribution with inattentive consumers by Daria Finocchiaro	2010:243
Identifying VARs through Heterogeneity: An Application to Bank Runs by Ferre De Graeve and Alexei Karas	2010:244
Modeling Conditional Densities Using Finite Smooth Mixtures by Feng Li, Mattias Villani and Robert Kohn	2010:245
The Output Gap, the Labor Wedge, and the Dynamic Behavior of Hours by Luca Sala, Ulf Söderström and Antonella Trigari	2010:246
Density-Conditional Forecasts in Dynamic Multivariate Models by Michael K. Andersson, Stefan Palmqvist and Daniel F. Waggoner	2010:247
Anticipated Alternative Policy-Rate Paths in Policy Simulations by Stefan Laséen and Lars E. O. Svensson	2010:248
MOSES: Model of Swedish Economic Studies by Gunnar Bårdsen, Ard den Reijer, Patrik Jonasson and Ragnar Nymoen	2011:249
The Effects of Endogenuos Firm Exit on Business Cycle Dynamics and Optimal Fiscal Policy by Lauri Vilmi	2011:250
Parameter Identification in a Estimated New Keynesian Open Economy Model by Malin Adolfson and Jesper Lindé	2011:251
Up for count? Central bank words and financial stress by Marianna Blix Grimaldi	2011:252
Wage Adjustment and Productivity Shocks by Mikael Carlsson, Julián Messina and Oskar Nordström Skans	2011:253

Stylized (Arte) Facts on Sectoral Inflation by Ferre De Graeve and Karl Walentin	2011:254
Hedging Labor Income Risk by Sebastien Betermier, Thomas Jansson, Christine A. Parlour and Johan Walden	2011:255
Taking the Twists into Account: Predicting Firm Bankruptcy Risk with Splines of Financial Ratios by Paolo Giordani, Tor Jacobson, Erik von Schedvin and Mattias Villani	2011:256
Collateralization, Bank Loan Rates and Monitoring: Evidence from a Natural Experiment by Geraldo Cerqueiro, Steven Ongena and Kasper Roszbach	2012:257
On the Non-Exclusivity of Loan Contracts: An Empirical Investigation by Hans Degryse, Vasso Ioannidou and Erik von Schedvin	2012:258
Labor-Market Frictions and Optimal Inflation by Mikael Carlsson and Andreas Westermark	2012:259
Output Gaps and Robust Monetary Policy Rules by Roberto M. Billi	2012:260
The Information Content of Central Bank Minutes by Mikael Apel and Marianna Blix Grimaldi	2012:261
The Cost of Consumer Payments in Sweden	2012:262
by Björn Segendorf and Thomas Jansson	
Trade Credit and the Propagation of Corporate Failure: An Empirical Analysis	2012:263
by Tor Jacobson and Erik von Schedvin	
Structural and Cyclical Forces in the Labor Market During the Great Recession: Cross-Country Evidence	2012:264
by Luca Sala, Ulf Söderström and AntonellaTrigari	
Pension Wealth and Household Savings in Europe: Evidence from SHARELIFE	2013:265
by Rob Alessie, Viola Angelini and Peter van Santen	
Long-Term Relationship Bargaining	2013:266
by Andreas Westermark	
Using Financial Markets To Estimate the Macro Effects of Monetary Policy: An Impact-Identified FAVAR*	2013:267
by Stefan Pitschner	
DYNAMIC MIXTURE-OF-EXPERTS MODELS FOR LONGITUDINAL AND DISCRETE-TIME SURVIVAL DATA	2013:268
by Matias Quiroz and Mattias Villani	
Conditional euro area sovereign default risk	2013:269
by André Lucas, Bernd Schwaab and Xin Zhang	
Nominal GDP Targeting and the Zero Lower Bound: Should We Abandon Inflation Targeting?*	2013:270
by Roberto M. Billi	
Un-truncating VARs*	2013:271
by Ferre De Graeve and Andreas Westermark	
Housing Choices and Labor Income Risk	2013:272
by Thomas Jansson	
Identifying Fiscal Inflation*	2013:273
by Ferre De Graeve and Virginia Queijo von Heideken	
On the Redistributive Effects of Inflation: an International Perspective*	2013:274
by Paola Boel	
Business Cycle Implications of Mortgage Spreads*	2013:275
by Karl Walentin	
Approximate dynamic programming with post-decision states as a solution method for dynamic	2013:276
	0040.077
A detrimental reedback loop: deleveraging and adverse selection	2013:277
Distortionany Ficeral Policy and Manatany Policy Coole	2012:270
by Klaus Adam and Roberto M. Billi	2013.218
by Maus Audin and Roberto W. Dilli Dredicting the Spread of Einancial Innovations: An Enidemiclogical Approach	2012-270
hy Isaiah Hull	2013.219
Firm-Level Evidence of Shifts in the Sunnly of Credit	2012.280
by Karolina Holmberg	2010.200

Lines of Credit and Investment: Firm-Level Evidence of Real Effects of the Financial Crisis by Karolina Holmberg	2013:281
A wake-up call: information contagion and strategic uncertainty by Toni Ahnert and Christoph Bertsch	2013:282
Debt Dynamics and Monetary Policy: A Note	2013:283
by Stefan Laséen and Ingvar Strid	
Optimal taxation with home production	2014:284
by Conny Olovsson	
Incompatible European Partners? Cultural Predispositions and Household Financial Behavior	2014:285
by Michael Haliassos, Thomas Jansson and Yigitcan Karabulut	
How Subprime Borrowers and Mortgage Brokers Shared the Piecial Behavior	2014:286
by Antje Berndt, Burton Hollifield and Patrik Sandås	
The Macro-Financial Implications of House Price-Indexed Mortgage Contracts	2014:287
by Isaiah Hull	
Does Trading Anonymously Enhance Liquidity?	2014:288
by Patrick J. Dennis and Patrik Sandås	
Systematic bailout guarantees and tacit coordination	2014:289
by Christoph Bertsch, Claudio Calcagno and Mark Le Quement	
Selection Effects in Producer-Price Setting	2014:290
by Mikael Carlsson	
Dynamic Demand Adjustment and Exchange Rate Volatility	2014:291
by Vesna Corbo	
Forward Guidance and Long Term Interest Rates: Inspecting the Mechanism	2014:292
by Ferre De Graeve, Pelin Ilbas & Raf Wouters	
Firm-Level Shocks and Labor Adjustments	2014:293
by Mikael Carlsson, Julián Messina and Oskar Nordström Skans	
A wake-up call theory of contagion	2015:294
by Toni Ahnert and Christoph Bertsch	
Risks in macroeconomic fundamentals and excess bond returns predictability	2015:295
by Rafael B. De Rezende	
The Importance of Reallocation for Productivity Growth: Evidence from European and US Banking	2015:296
by Jaap W.B. Bos and Peter C. van Santen	
SPEEDING UP MCMC BY EFFICIENT DATA SUBSAMPLING	2015:297
by Matias Quiroz, Mattias Villani and Robert Kohn	
Amortization Requirements and Household Indebtedness: An Application to Swedish-Style Mortgages	2015:298
by Isaiah Hull	
Fuel for Economic Growth?	2015:299
by Johan Gars and Conny Olovsson	
Searching for Information	2015:300
by Jungsuk Han and Francesco Sangiorgi	
What Broke First? Characterizing Sources of Structural Change Prior to the Great Recession	2015:301
by Isaiah Hull	
Price Level Targeting and Risk Management	2015:302
by Roberto Billi	
Central bank policy paths and market forward rates: A simple model	2015:303
by Ferre De Graeve and Jens Iversen	
Jump-Starting the Euro Area Recovery: Would a Rise in Core Fiscal Spending Help the Periphery?	2015:304
by Olivier Blanchard, Christopher J. Erceg and Jesper Lindé	
Bringing Financial Stability into Monetary Policy*	2015:305
by Eric M. Leeper and James M. Nason	
SCALABLE MCMC FOR LARGE DATA PROBLEMS USING DATA SUBSAMPLING AND THE DIFFERENCE ESTIMATOR	2015:306
by MATIAS QUIROZ, MATTIAS VILLANI AND ROBERT KOHN	

SPEEDING UP MCMC BY DELAYED ACCEPTANCE AND DATA SUBSAMPLING by MATIAS QUIROZ	2015:307
Modeling financial sector joint tail risk in the euro area by André Lucas, Bernd Schwaab and Xin Zhang	2015:308
Score Driven Exponentially Weighted Moving Averages and Value-at-Risk Forecasting	2015:309
by André Lucas and Xin Zhang	
On the Theoretical Efficacy of Quantitative Easing at the Zero Lower Bound	2015:310
by Paola Boel and Christopher J. Waller	
Optimal Inflation with Corporate Taxation and Financial Constraints	2015:311
by Daria Finocchiaro, Giovanni Lombardo, Caterina Mendicino and Philippe Weil	2010.011
Fire Sale Bank Recapitalizations	2015:312
by Christoph Bertsch and Mike Mariathasan	2010.012
Since you're so rich, you must be really smart: Talent and the Finance Wage Premium	2015:313
by Michael Böhm, Daniel Metzger and Per Strömberg	2010.010
Debt. equity and the equity price puzzle	2015:314
by Daria Finocchiaro and Caterina Mendicino	2010.011
Trade Credit: Contract-Level Evidence Contradicts Current Theories	2016:315
by Tore Ellingsen, Tor Jacobson and Frik von Schedvin	2010.010
Double Liability in a Branch Banking System: Historical Evidence from Canada	2016:316
by Anna Grodecka and Antonis Kotidis	2010.010
Subprime Borrowers, Securitization and the Transmission of Business Cycles	2016:317
by Anna Grodecka	2010.011
Real-Time Forecasting for Monetary Policy Analysis: The Case of Sveriges Riksbank	2016:318
by Jens Iversen. Stefan Laséen, Henrik Lundvall and Llf Söderström	2010.010
Fed Liftoff and Subprime Loan Interest Rates: Evidence from the Peer-to-Peer Lending	2016:319
by Christoph Bertsch, Isaiah Hull and Xin Zhang	2010.010
Curbing Shocks to Corporate Liquidity: The Role of Trade Credit	2016:320
by Niklas Amberg Tor Jacobson Frik von Schedvin and Robert Townsend	2010.020
Firms' Strategic Choice of Loan Delinguencies	2016:321
by Paola Morales-Acevedo	2010.021
Fiscal Consolidation Under Imperfect Credibility	2016:322
by Matthieu Lemoine and Jesper Lindé	2010.022
Challenges for Central Banks' Macro Models	2016:323
by Jesper Lindé Frank Smets and Rafael Wouters	2010.020
The interest rate effects of government bond purchases away from the lower bound	2016:324
by Rafael B. De Rezende	2010.024
COVENANT-LIGHT CONTRACTS AND CREDITOR COORDINATION	2016:325
by Bo Becker and Victoria Ivashina	2010.020
Endogenous Separations Wage Rigidities and Employment Volatility	2016:326
by Mikael Carlsson and Andreas Westermark	2010.020
Renovatio Monetae: Gesell Taxes in Practice	2016:327
by Roger Svensson and Andreas Westermark	2010.021
Adjusting for Information Content when Comparing Forecast Performance	2016:328
by Michael K. Andersson. Ted Aranki and André Reslow	2010.020
Economic Scarcity and Consumers' Credit Choice	2016:329
by Marieke Bos. Chloé Le Cog and Peter van Santen	
Uncertain pension income and household saving	2016:330
by Peter van Santen	
Money, Credit and Banking and the Cost of Financial Activity	2016:331
by Paola Boel and Gabriele Camera	2010.001
Oil prices in a real-business-cycle model with precautionary demand for oil	2016:332
by Conny Olovsson	2010.002
Financial Literacy Externalities	2016:333
by Michael Haliasso, Thomas Jansson and Yigitcan Karabulut	2010.000

The timing of uncertainty shocks in a small open economy by Hanna Armelius, Isaiah Hull and Hanna Stenbacka Köhler	2016:334
Quantitative easing and the price-liquidity trade-off by Marien Ferdinandusse, Maximilian Freier and Annukka Ristiniemi	2017:335
What Broker Charges Reveal about Mortgage Credit Risk	2017:336
	0017.007
Asymmetric Macro-Financial Spillovers	2017:337
by Kristina Bluwstein Leteney Arbitrage When Marketa Become Fester	2017.220
by Burton Hollifield, Batrik Sandås and Andrew Todd	2017.330
How big is the toolbox of a central banker? Managing expectations with policy-rate forecasts: Evidence from Sweden	2017:339
by Magnus Åhl	
International business cycles: quantifying the effects of a world market for oil	2017:340
by Johan Gars and Conny Olovsson I	
Systemic Risk: A New Trade-Off for Monetary Policy?	2017:341
by Stefan Laséen, Andrea Pescatori and Jarkko Turunen	
Household Debt and Monetary Policy: Revealing the Cash-Flow Channel	2017:342
by Martin Flodén, Matilda Kilström, Jósef Sigurdsson and Roine Vestman	
House Prices, Home Equity, and Personal Debt Composition	2017:343
by Jieving Li and Xin Zhang	
Identification and Estimation issues in Exponential Smooth Transition Autoregressive Models	2017:344
by Daniel Buncic	
Domestic and External Sovereign Debt	2017:345
by Paola Di Casola and Spyridon Sichlimiris	
The Role of Trust in Online Lending by Christoph Bertsch, Isaiah Hull, Yingjie Qi and Xin Zhang	2017:346
On the effectiveness of loan-to-value regulation in a multiconstraint framework by Anna Grodecka	2017:347
Shock Propagation and Banking Structure by Mariassunta Giannetti and Farzad Saidi	2017:348
The Granular Origins of House Price Volatility	2017:349
by Isaiah Hull, Conny Olovsson, Karl Walentin and Andreas Westermark	
Should We Use Linearized Models To Calculate Fiscal Multipliers?	2017:350
by Jesper Lindé and Mathias Trabandt	
The impact of monetary policy on household borrowing – a high-frequency IV identification by Maria Sandström	2018:351
Conditional exchange rate pass-through: evidence from Sweden by Vesna Corbo and Paola Di Casola	2018:352
Learning on the Job and the Cost of Business Cycles by Karl Walentin and Andreas Westermark	2018:353
Trade Credit and Pricing: An Empirical Evaluation by Niklas Amberg, Tor Jacobson and Erik von Schedvin	2018:354
A shadow rate without a lower bound constraint by Rafael B. De Rezende and Annukka Ristiniemi	2018:355
Reduced "Border Effects", FTAs and International Trade by Sebastian Franco and Erik Frohm	2018:356
Spread the Word: International Spillovers from Central Bank Communication by Hanna Armelius, Christoph Bertsch, Isaiah Hull and Xin Zhang	2018:357
Predictors of Bank Distress: The 1907 Crisis in Sweden by Anna Grodecka, Seán Kenny and Anders Ögren	2018:358

Diversication Advantages During the Global Financial Crisis by Mats Levander	2018:359
Towards Technology-News-Driven Business Cycles by Paola Di Casola and Spyridon Sichlimiris	2018:360
The Housing Wealth Effect: Quasi-Experimental Evidence by Dany Kessel, Björn Tyrefors and Roine	2018:361
Identification Versus Misspecification in New Keynesian Monetary Policy Models by Malin Adolfson, Stefan Laseén, Jesper Lindé and Marco Ratto	2018:362
The Macroeconomic Effects of Trade Tariffs: Revisiting the Lerner Symmetry Result by Jesper Lindé and Andrea Pescatori	2019:363
Biased Forecasts to Affect Voting Decisions? The Brexit Case by Davide Cipullo and André Reslow	2019:364
The Interaction Between Fiscal and Monetary Policies: Evidence from Sweden by Sebastian Ankargren and Hovick Shahnazarian	2019:365
Designing a Simple Loss Function for Central Banks: Does a Dual Mandate Make Sense? by Davide Debortoli, Jinill Kim and Jesper Lindé	2019:366
Gains from Wage Flexibility and the Zero Lower Bound by Roberto M. Billi and Jordi Galí	2019:367
Fixed Wage Contracts and Monetary Non-Neutrality by Maria Björklund, Mikael Carlsson and Oskar Nordström Skans	2019:368
The Consequences of Uncertainty: Climate Sensitivity and Economic Sensitivity to the Climate by John Hassler, Per Krusell and Conny Olovsson	2019:369
Does Inflation Targeting Reduce the Dispersion of Price Setters' Inflation Expectations? by Charlotte Paulie	2019:370
Subsampling Sequential Monte Carlo for Static Bayesian Models by David Gunawan, Khue-Dung Dang, Matias Quiroz, Robert Kohn and Minh-Ngoc Tran	2019:371
Hamiltonian Monte Carlo with Energy Conserving Subsampling by Khue-Dung Dang, Matias Quiroz, Robert Kohn, Minh-Ngoc Tran and Mattias Villani	2019:372
Institutional Investors and Corporate Investment by Cristina Cella	2019:373
The Impact of Local Taxes and Public Services on Property Values by Anna Grodecka and Isaiah Hull	2019:374
Directed technical change as a response to natural-resource scarcity by John Hassler, Per Krusell and Conny Olovsson	2019:375
A Tale of Two Countries: Cash Demand in Canada and Sweden by Walter Engert, Ben Fung and Björn Segendorf	2019:376
Tax and spending shocks in the open economy: are the deficits twins? by Mathias Klein and Ludger Linnemann	2019:377
Mind the gap! Stylized dynamic facts and structural models by Fabio Canova and Filippo Ferroni	2019:378
Financial Buffers, Unemployment Duration and Replacement Labor Income by Mats Levander	2019:379
Inefficient Use of Competitors' Forecasts? by André Reslow	2019:380
How Much Information Do Monetary Policy Committees Disclose? Evidence from the FOMC's Minutes and Transcripts by Mikael Apel, Marianna Blix Grimaldi and Isaiah Hull	2019:381
Risk endogeneity at the lender/investor-of-last-resort by Diego Caballero, André Lucas, Bernd Schwaab and Xin Zhang	2019:382
Heterogeneity in Households' Expectations of Housing Prices – Evidence from Micro Data by Erik Hjalmarsson and Pär Österholm	2019:383
Big Broad Banks: How Does Cross-Selling A Affect Lending? by Yingjie Qi	2020:384
Unemployment Fluctuations and Nominal GDP Targeting by Roberto Billi	2020:385



Sveriges Riksbank Visiting address: Brunkebergs torg 11 Mail address: se-103 37 Stockholm

Website: www.riksbank.se Telephone: +46 8 787 00 00, Fax: +46 8 21 05 31 E-mail: registratorn@riksbank.se